AN ETHNOGRAPHIC STUDY OF FACTORS THAT ELEMENTARY SCHOOL PRINCIPALS PERCEIVE CONTRIBUTE TO INCREASED GRADE SIX MEAN COMPOSITE SCORES ON THE <u>CANADIAN TESTS OF BASIC SKILLS</u> OVER A SIX YEAR PERIOD

CENTRE FOR NEWFOUNDLAND STUDIES TOTAL OF 10 PAGES ONLY MAY BE XEROXED

GERALD NEIL JAMES MERCER







AN ETHNOGRAPHIC STUDY OF FACTORS THAT ELEMENTARY SCHOOL PRINCIPALS PERCEIVE CONTRIBUTE TO INCREASED GRADE SIX MEAN COMPOSITE SCORES ON THE CANADIAN TESTS OF BASIC SKILLS

OVER A SIX YEAR PERIOD

BY

© GERALD NEIL JAMES MERCER, B.A.(Ed.)

A thesis submitted to the School of Graduate Studies

in partial fulfilment of the requirements for the degree of

Master of Education

Faculty of Education

Memorial University of Newfoundland

1993

St. John's

Newfoundland



National Library of Canada Bibliothèque nationale du Canada

Acquisitions and Bibliographic Services Branch Direction des acquisitions et des services bibliographiques

395 Mellington Street Ottawa, Ontario K1A 0N4 395, tue Welington Ottav. a (Ontano) K1A ON4

Net for Servicese

iter services

The author has granted an irrevocable non-exclusive licence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of his/her thesis by any means and in any form or format, making this thesis available to interested persons. L'auteur a accordé une licence irrévocable et non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter, distribuer ou vendre des copies de sa thèse de quelque manière et sous quelque forme que ce soit pour mettre des exemplaires de cette thèse à la disposition des personnes intéressées.

The author retains ownership of the copyright in his/her thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without his/her permission. L'auteur conserve la propriété du droit d'auteur qui protège sa thèse. Ni la thèse ni des extraits substantiels de celle-ci ne doivent être imprimés ou autrement reproduits sans son autorisation.

ISBN 0-315-86630-6

Canadä

ABSTRACT

The Government of Newtoundland Department of Education has administered the <u>Canadian Tests of Basic Skills</u> to grade six students every three years since 1976. Newtoundland schools traditionally score below Canadian norms on these tests and strive to improve by implementing revised policies, guidelines, and up-dated curricula published by the Department of Education. Yet, only 31 schools in the province have succeeded in registering a mean composite score increase between 1985 and 1988 and again between 1988 and 1991. Following case study methodology reported by Yin (1984) and Merriam (1988), an exploratory study was undertaken to identify common factors that school principals perceive have contributed to the grade six increased mean composite scores.

During May and June 1992, a survey was distributed to 31 school principals. Seventy-four percent of the surveys were returned. The analysis of the data revealed five distinct themes: <u>Student Attitudes toward School</u>: <u>Innovation and Planned Change</u>; <u>Resource-based Learning as an Innovation</u>; <u>Public Perceptions of School Life</u>; and <u>Public Support and Principal</u> <u>Accessibility</u>. These themes were explored further in four case study interviews.

A sample of four survey respondents was identified using a purposive

sampling method (Merriam, 1988). In February and March 1993, four focused interviews (Yin, 1984) were conducted and revealed five themes reflecting common principal perceptions and/or school practices. The four schools:

- implemented interventions with an aim to foster positive student attitudes toward school;
- implemented innovations that reflect current educational thought and respond to needs identified by teachers and administrators working together;
- have support for school-wide innovations from educators and members of the community;
- serve parents who support their children's educational endeavours and a public that feels school is beneficial for children; and
- have a staff that is committed to the educational welfare of students and to personal professional development.

The researcher recommends that future studies continue to explore this phenomenon.

ACKNOWLEDGEMENTS

I would like to express my sincere appreciation for the support, understanding, and guidance given to me by professors, colleagues, and family members.

To Dr. G. Koski, a respected advisor, I extend a hand of appreciation and thanks for his encouragement, patience, and guidance.

To the principals who gave of their valuable time to participate in this study, I extend a sincere thanks and congratulations for their leadership and visions that impact positively on the students in their schools.

To my colleagues employed by the Conception Bay South Integrated School Board, I extend my gratitude for recognizing the value of this research.

To my wife, Daphne whose patience, understanding, commitment, encouragement and love supported and sustained me throughout my graduate studies, I extend a special thanks and my love.

To my two wonderful children, Rebecca and Andrew, born during the years of my graduate studies, I promise not to go to work so much.

iv

TABLE OF CONTENTS

ABS	TRACT ii
ACK	NOWLEDGEMENTS iv
LIST	OF TABLES
СНА	PTER
1.	NATURE OF THE STUDY 1
	Introduction
	Background to the Problem 1
	Significance of the Study 6
	Limitations of the Study6
	Summary
2.	REVIEW OF RELATED LITERATURE
	The Canadian Tests of Basic Skills
	An Historical Perspective
	Historical Perspective of Effective Schools Literature
	Introduction 15 The Search for Effective Schools 15 The Search for Determinants of Effective Schools 18 Effective vs. Successful Schools 20 Effective vs. Efficient Schools 20 Effective School Attributes 21 The Value of the Effective Schools 20 Effective vs. Efficient Schools 20 Effective vs. Efficient Schools 20 Effective vs. Efficient Schools 21 The Value of the Effective Schools Movement 25

	School Improvement	
	from Planned Change to Educational Reform	9
	Introduction	29
	Innovation and Planned Change	10
	Planned Social and Educational Change	31
	The Process of Planned Educational Change	38
	The Players in the Process and their Roles	14
	Converging Literatures	57
	School Improvement	58
	Reform: A Call for Restructuring the Education System	57
	Summary	68
3.	METHODOLOGY	70
	Research Design	70
	Research Phases	70
	The First Research Question	70
	The Survey	71
	Development of the Instrument	71
	Administration of the Survey	73
	Data Analysis Procedures	74
	The Second Research Question	74
	The Case Study	75
	Case Study Propositions	75
	Unit of Analysis	76
	The Logic Linking the Data to the Propositions and Criteria	
	for Interpreting the Findings	76
	The Interview Strategy	78
	Sample Selection	79
	Administration of the Case Study	81

vi

	vii
4.	REPORT AND ANALYS/S OF RESULTS
	Introduction
	Organization of the Findings
	Phase One: Survey Information
	Part I: Community Changen
F	Phase Two: Case Study 88
5	The Four Schools 91 Student Attitudes Toward School 94 Innovation and Planned Change 98 Resource-based Learning as an Innovation 101 Public Perceptions of School Life 105 Public Support and Principal Accessibility 108 Other Factors 112
5.	CONCLUSION AND RECOMMENDATIONS 115
	Summary
	Conclusions
	Summary 119
	Recommendations
BIBI	LIOGRAPHY 121
APP	PENDIX A
APF	PENDIX B
APF	PENDIX C 137
APF	PENDIX D

LIST OF TABLES

Table 1 Percentile Ranks for Grade Six Students in Newfoundland 1976-1991	13
Table 2 Percentage of Respondents Agreeing and Disagreeing with Statements	86
Table 3 Percentage of Respondents Agreeing and Disagreeing with Statements	APPENDIX B - 134
Table 4 Percentage of Respondents Agreeing and Disagreeing with Statements	. APPENDIX C - 137
Table 5 Grade VI Mean Composite Scores on CTBS and Increase Organized by School	89
Table 6 Student Populations	
Table 7 Principals' Qualifications and Employment Data .	91

CHAPTER ONE

Nature of the Study

Introduction

The purpose of this study was to explore the perceptions of elementary school principals as they relate to factors that contributed to an increase in grade six composite scores on the <u>Canadian Tests of Basic Skills</u> (hereafter referred to as CTBS) over a six-year period.

Background to the Problem

In the Newfoundiand and Labrador Government Department of Education publication entitled, <u>The Aims of Public Education</u> (1984), there exists a list of objectives that translates abstract aims of education into objectives that guide expected practice. Objective six states, "to ensure that all pupils master the fundamental skills of learning to the limit of their abilities" (p. 6). These fundamental skills of learning can be referred to as "Jasic skills". Schools in Newfoundiand strive to build learning environments that are conducive to mastering these basic skills.

To achieve these aims, the Newfoundland Government Department of Education provides support and guidance for educators employed by school districts. Curriculum and policy guides are written and distributed in an effort to improve the means by which school programs are delivered. These documents are based on current research, Department of Education policies, and educational needs in the province of Newfoundland.

One such publication entitled, <u>Learning to Learn</u>, was authorized by the Minister of Education in 1991. The document contains "policies and guidelines for the implementation of resource-based learning in Newfoundland and Labrador Schools" (cover). The 1990 publication entitled, <u>Early Childhood</u> <u>Program Guide</u> is another that reflects current research and educational policy for classroom, practice in primary schools. By publishing documents such as these, the Department of Education attempts to provide guidance for the instruction in the province's classrooms, support its educators, and ensure that students learn the skills that enable them to become independent, lifelong learners.

In an effort to assess one aspect of the effectiveness of provincial schools, the Department of Education administers the CTBS to students in grades four, six, and eight - one grade level on a rotating basis, annually. In October, each student completes the age-appropriate battery of tests. Student response sheets are submitted to the Department of Education for computer and statistical analysis. The resulting data are analyzed and reported in two ways:

2

- specific student and district data are disseminated to respective school boards for discussion; and
- a government document entitled, <u>Testing Standards</u>, is published and distributed to the public, the intent of which is to present and discuss provincial results.

The Government of Newfoundiand and Labrador has provided schools and school districts with direction for improved curriculum delivery, up-dated or new policies, and regular reports of basic skill achievement, yet the basic skills of Newfoundiand students, as measured on the CTBS remain unacceptable to the Department of Education and to the public whose tax dollars support this system. Generally, recent composite scores on the CTBS have been below Canadian norms.

In the 1991 administration of the Canadian Tests of Basic Skills (CTBS), our Grade 6 students placed at the 37th percentile in overall achievement. This means that the average performance was better than 37% of the students in the norm group. (Government of Newtoundiand, 1991, p.1)

The Government of Newfoundland and Labrador has recognized the need to improve the effectiveness of schools in the province. In January 1990, it published the report of the School Improvement/ Effectiveness Committee entitled. <u>Challenge for Excellence</u>. This report acknowledges the challenge of effecting improvement in our province's schools:

In Newfoundland and Labrador, as elsewhere, schools differ in the

degree to which they meet the real needs of students. Some would say that our schools vary in their level of excellence, while other more critical observers question their basic effectiveness. Perhaps the most relevant observation is that our schools vary significantly in their commitment to narrowing the gap between what they should be and what they are. In effect, our schools vary in their commitment to improvement initiatives which are designed to enhance their overall effectiveness in educating our youth. (pp. 1-2)

This document attempts to provide appropriate rationale and a plan of action to improve the effectiveness of schools in Newfoundland.

The product of millions of tax-payer dollars is disenchanting. The

Department of Education has attempted to provide guidance for improved

curriculum delivery and a means for tracking student basic skill achievement,

yet most schools in our province have failed to provide programs and

instruction that fosier improved basic skill performance.

There exists a positive corollary to this dismal situation. In this province, a relatively small number of schools have effected a positive change in the mean composite scores on the grade six CTBS over a six year period. This phenomenon forms the basis of this study. It would be beneficial to explore the factors that have contributed to the positive basic skill increases as measured by CTBS

Newfoundland students in grade six wrote the CTBS in 1985, 1988, and 1991. To identify the group that has demonstrated mean composite score improvement over the six years of testing, the author requested and received from the Government of Newfoundland Department of Education appropriate data sorted by school, year of testing, and number of students participating in the third year, 1991. These data were analyzed to identify schools that reflected the following two criteria:

1. the 1991 grade six population was greater than five students; and

 the mean composite scores between 1985 and 1988 and again between 1988 and 1991 increased.

Of the original data, the number of schools reporting five or more students writing the CTBS was 288. Of these schools, 31 schools met the population and improvement criteria delineated for the study. Analysis of the circumstances and/or factors that lead to school improvement may reveal information that could enhance school performance.

The school principal is the person who theoretically is responsible for planning, leading, organizing and controlling school life (Sergiovanni, 1991, pp. 17-21). The principal guides and balances response to changing curriculum, use of funds, teacher deployment, community liaison, and other pertinent aspects of administering the school environment. This person is the one who is in control of and is in tune with the factors that affect the daily performance of students.

An increase in the mean composite scores over time suggests that some factors or interventions, whether passive or active, have contributed to bringing

5

about the increase. The identification of factors common to diverse school environments could be beneficial to other visionary educators who would like an insight into other schools that have successfully fostered an increase in student basic skill competencies.

Significance of the Study

Hakim (1987) states, "Qualitative research may be used for preliminary exploratory work before mounting a larger scale or more complex study" (p. 28). The intent of this study was to use qualitative research methodology to explore principals' perceptions of factors that have contributed to increases in the grade six mean composite scores on the CTBS over a six-year period. No study was found that has explored this phenomenon. These findings could form a basis for initiating a larger, more comprehensive study.

Limitations of the Study

The following limitations were recognized in the planning of this study:

 The data collected through the administration of the CTBS is limited to the basic skills tested by the standardized test.

2. This study explores the perceptions of the school principal and

therefore the findings are limited to that group of educators.

 The number of principals involved in these case studies was four and conclusions can be made in relation only to those who participated.

4. The survey instrument, usually a quantitative research tool, served as a focus for the design of the case study interviews. Given the small number of schools that displayed a grade six mean composite score increase over a six-year period, the value of designing a quantitative study was questionable. Therefore, given that a control group was not identified, quantitative statistical analysis could not be applied to the data collected from the survey instrument.

Summary

This thesis reports the findings of a qualitative research study conducted in the spring of 1992 and the winter of 1993. It identified school principals' common perceptions of factors that contributed to grade six mean composite score increases on the CTBS between 1985 and 1991 in the province of Newloundland.

Chapter Two of this thesis presents an overview of the history and development of the CTBS, effective schools literature, and planned educational change literature.

Chapter Three reports the methodology and the procedures followed to

conjulete the case studies of four elementary school principals.

Chapter Four presents, in summary form, the data collected in each phase of the research and an analysis of that data.

Chapter Five reports the findings of the study, and a list of

recommendations for further research projects.

CHAPTER TWO

Review of Related Literature

The Canadian Tests of Basic Skills

An Historical Perspective

The <u>Canadian Tests of Basic Skills</u> (hereafter referred to as CTBS) is a battery of tests that measures educational growth and development in the areas of vocabulary, reading, language arts, work-study skills, and mathematics. Based on the <u>lowa Tests of Basic Skills</u> and first published in 1955 by the Nelson Publishing Company, the tests are intended to assess the generalized educational achievement of Canadian school children (Gallivan, 1986). Scores are interpreted using norms established using a standardization sample representing over 100 schools from every Canadian province and territory.

The CTBS has a history of content and standardization changes. The original version, published in 1955, included primary and elementary sections. An elementary battery was published in 1966. Content and standardization changes occurred in 1974 and 1982. A primary battery was added in 1972. Metric versions became available in 1976 and a high school version was added in 1981 (Galiwan, 1986).

The elementary version of the CTBS can be administered by classroom teachers. It takes five hours to complete the test in the classroom setting. It is recommended that the timed individual tests be scheduled in four to eight sessions over at least a four day period.

The test is divided into various levels. These levels allow students to begin reading at an appropriate point of the test determined by school grade. Grade six students begin at level 12. All test items are presented in multiple choice format. Students are instructed to read each test item and report responses by filling in a bubble on the answer sheet. The elementary battery of the CTBS is comprised of eleven subtests: one vocabulary subtest that measures isolated and contextual word comprehension; one reading comprehension subtest that measures picture, sentence, and story comprehension skills; four language skills subtests that measure spelling, capitalization, punctuation, and usage ability; two work study skills subtests referred to as W-1 that measures map, graph, and table interpretation skills, and W-2 that measures reference materials utilization skills: and three mathematics subtests referred to as M-1 that measures the knowledge of the number system and basic arithmetical terms and operations, M-2 that measures the ability to solve problems presented in verbal form, and M-3 that measures the ability to solve problems presented out of context.

The tests are scored for each student either mechanically or by hand. Norms based on a national standardization sample selected from over 100 schools in every Canadian province and territory are included with the testing

10

package. Individual student scores can be reported subtest-by-subtest in the form of a raw score, a grade level equivalency, and a national percentile. Composite scores derived from the eleven subtests can be calculated for each student.

The Newfoundland Grade Six Perspective

The Government of Newfoundland Department of Education publishes a summary of findings entitled, <u>Testing Standards</u> in each year that the CTBS are administered. The following discussion is based on the Department of Education's perspective as reported by the 1988 and 1991 editions.

Grade six testing years.

Newfoundland grade six students were tested using the CTBS in 1976, 1979, 1982, 1985, 1988, and 1991.

Reporting composite scores.

The mean composite score for grade six students in Newfoundland schools is reported in two distinct forms: a grade equivalent and a national percentile rank. "When the tests were normed in October, 1980, the score at the 50th percentile (the median score) in Grade 5 was assigned the gradeequivalent score of 52.0, the median score in the grade 6 group was assigned

11

the value of 62.0" (Government of Newfoundiand Testing Standards, 1988, p. 3). Grade equivalent is reported somewhat differently in the 1991 document. "Grade equivalent scores indicate the grade level at which the typical students make the raw score. For example, if a student is assigned a grade equivalent of 6.2, it means that the number of items (s)he has correct is equal to that which an average student in the second month (October) of Grade 6 would have correct" (p. 3).

The 1988 data.

The actual scores reported in the document for Newfoundland students are 60.0 (Grade Equivalent) and 45 (National Percentile Rank). These scores translate into a slightly below average performance. The document explains the implications of this analysis: "If each student in Newfoundland had had one more item correct on each subtest, the provincial mean composite score would have been at the national norm" (p. 5).

The 1991 data.

The actual scores reported in the document for Newfoundland students are 5.8 (Grade Equivalent) and 37 (National Percentile Rank). The document suggests that these scores translate into a below average range when average is defined as 40th to 60th percentile. The document explains, "On the complete battery of test [sic], however, the average performance of Newfoundland students would have had to increase by 20 items for our performance to approximate the national norm^{*} (p. 5).

General trends in composite scores from 1976 - 1991.

Table 1 presents the composite scores, reported as percentile ranks, from each of the grade six testing ,ears.

Table 1

Percentile Ranks for Grade Six Students in Newfoundland 1976-1991

Composite Scores	36	45	41	46	45	37

Note. These scores are taken from the 1991 and 1988 editions of the Government of Newfoundland Department of Education publications entitled, <u>Testing Standards</u>.

A distinct trend can be seen. Toward the late 1970s, there was a general improvement in the mean composite scores derived from the data collected trom grade six students. This trend levelled off in the 1980s. In 1991, the mean composite score declined significantly. In the 1980s Newfoundland st. dents had not improved in their general basic skills competencies as tested by CTB^{C.} In lact, students writing the CTBS in 1991 were registering below average mean composite scores when compared to national norms. In the summary of the 1991 document entitled, <u>Testing Standards</u>, there exist statements which report the Government of Newfoundland's perspective on the phenomenon. First, there are criticisms of the CTBS "Canadian Tests of Basic Skills are frequently criticized because they do not take into account the processes that students go through as they learn" (p. 45). "The language subtests are criticized because they do not measure how well students can write" (p. 45). "From a mathematics perspective, the criticism focuses on the inclusion of operations with fraction [sic] which receive no emphasis in our primary and elementary programs" (p. 45). The author of the document counters each of the criticisms but the last. It is acknowledged that the difficulties identified in the mathematics portion of the CTBS must be addressed if improvements are to be seen.

The final paragraph of the summary reflects the general view of the use of CTBS in the province of Newfoundland and Labrador:

Although the acquisition of basic skills in [sig] an important objective of the school program, there are many other important objectives, such as moral and spiritual development that are not easily measured. No attempt is made to measure progress toward the achievement of these objectives. Consequently, the Canadian Tests of Basic Skills can be regarded as an important but not a complete measure of the productivity of the Newtoundand school system. (pp. 45-46)

Historical Perspective of Effective Schools Literature

Introduction

Ronald Edmonds (1982) wrote, "Educators have become increasingly convinced that the characteristics of schools are important determinants of academic achievement" (p.4). This is an opposing view to those held in the 1960s and early 1970s when the Coleman Report (1966) sent shock waves through the educational communities of the United States.

The Search for Effective Schools

The search for ways to build effective schools is decades old and has taken place not only in North America, but in Europe and Australia. Holmes (1989) identified sources of effective school research that date back to the 1930s and 1940s when specific projects suggested practices that could lead to school effectiveness. The launch of Sputnik in 1957 incide a plethora of criticism of the educational system in the United States for it was generally believed that the U.S.S.R. was quickly becoming technologically superior (Nicholls, 1983, p. 10). The public felt that the educational system was to blame; it was not as effective as it should have been.

Educational researchers began to pursue a magical recipe that would make every school in North America effective. Purkey and Smith (1983) cautioned against this. They recognized that there was no simple recipe or an easy-to-assemble model that would produce an effective school. Descriptive lists of characteristics of effective schools continued to emerge. Sergiovanni (1991) comments on the proper use of this literature:

Indiscriminate application of school-effectiveness research findings and, in particular, the development of generic lists of correlates or indicators that are subsequently applied uniformly to schools pose serious questions about the proper use of research and can result in negative, unanticicated consequences for teaching and learning. (0, 91)

The direction that the literature took was descriptive, yet there remained an underlying desire to create more effective schools. The reasons for this can be traced back to the mid 1960s.

The catalyst for a vast amount of the most recent effective schools research was the 1966 J. S. Coleman, E. Campbell, C. Hobson, J. McPartland, A. Mood, F. Weinfield, and R. York report of a survey of educational opportunity in the United States of America entitled, <u>Equality of</u> <u>Educational Opportunity</u> (for example see Block, 1983, p. 15; Robinson, 1985, pp. 1-3). This report, referred to in the literature as "The Coleman Report" after its primary author James S. Coleman, was written in response to section 402 of the United States of America Civil Rights Act of 1964. The act specifically called for a survey "concerning the lack of availability of equal educational opportunities for all individuals by reason of race, color, religion, or national origin in public educational institutions". A conclusion cited in the Coleman report states that:

schools bring little influence to bear on a child's achievement that is independent of his background and general social context; and that this very lack of an independent effect means that the inequalities imposed on children by their home, neighborhood, and peer environment are carried along to become the inequalities with which they confront adult life at the end of school. (p. 325)

This conclusion dealt a severe blow to the educational establishment for it suggested that schools, teachers, and the amount of money spent on education had little impact on the academic achievement of its students. Instead, socioeconomic backgrounds, location of school facilities, peer influence, and racial ethnicity - all factors that could not be controlled within the educational hierarchy - determined the student's rate of success (Davis & Thomas, 1989; Robinson, 1985; Block, 1983).

The Coleman Report had devastating effects on the educational community. Since it found that the school had little impact on student success, then academic excellence had to be determined by factors that could not be manipulated by educators. This led to lowered expectations for students, despondency, and decreased confidence in the importance of public education (Robinson, 1985).

During the years that followed the release of the Coleman report, critics of Coleman's methodology emerged. Block's 1983 summary of effective schools research reported a synthesis of these critiques citing shortcomings in the areas of survey response rate, validity of measures, the cross-sectional nature of the survey, use of school and district averages, and statistical procedures.

Still, there were schools whose lower socioeconomic populations were performing at higher academic levels (Sizer, 1985). The question followed, "Were these schools aberrations or were there other factors that helped determine academic excellence and ultimately successful schools?"

The Search for Determinants of Effective Schools

Researchers who investigated characteristics of effective schools traditionally studied schools whose student populations performed significantly higher than the statistical average in academic endeavours and on basic skill standardized tests. Sergiovanni (1991) suggested that one reason for the popularity of using student outcomes on standardized tests was the ease of obtaining the data. Researchers obtained this data and compared outliers those schools whose student populations scored consistently lower than the statistical mean with those whose populations scored significantly higher than the statistical mean (Purkey & Smith, 1983). Other researchers used case study methodology to explore the characteristics of these outlier schools (Beare, Caldwell, & Millikan, 1989). Many studies compared highly effective and highly ineffective schools matched on socioeconomic criteria (Davis & Thomas, 1989). In any case, the determinant of an effective school was

18

unidimensional - academic achievement.

Purkey and Smith (1983) suggested that comparing effective and ineffective schools with average schools was more valuable than comparing outliers. Ineffective schools best benefit from knowing why they have been deemed ineffective and not average. By comparing outlier ineffective schools to outlier effective schools differences are highlighted. These differences, while heralding the effective school, may be unattainable for the ineffective school. Purkey and Smith suggested that it would be more beneficial for educators to identify factors that make ineffective and effective schools outliers.

The literature cites many critics of the research who define effectiveness by the unidimensional criterion of high academic achievement (see Sergiovanni, 1991; Beare et al., 1989). Educational philosophers warned of the return to an early 1900s styled curriculum where effective schools would provide instruction in areas deemed appropriate to effective schools; of university based programs that would focus on basic skills in reading, mathematics, and science while neglecting the arts; of fixed promotion policies; and of a focus on merely the academic (Beare et al., 1989). While there remained a consensus among members of the public that academic achievement was important, there were other factors that had to be considered when affixing to any school the effective school seal of approval.

Effective vs. Successful Schools

More recent authors have found it necessary to separate the notion of an effective school from that of a successful school. Sergiovanni (1991) states that the terms effective and successful "are often used interchangeably to describe the same school or to communicate the same level of accomplishment" (p. 76). The two, he suggests, are distinctly different. Sergiovanni describes the accepted notion of an effective school: "An effective school is understood to be a school whose students achieve well in basic skills as measured by achievement tests." (p. 76). The term successful schools communicates a broader, more comprehensive view of effectiveness. It includes a multi-dimensional approach to describing high quality schooling.

Effective vs. Efficient Schools

Before one can consider the features of an effective school, one must be cognizant of the school environment and its resources. It could be hypothesized that the ideal school - one with unlimited resources, highly trained teachers, low student/teacher ratios, state-of-the-art equipment and facilities, and a supportive community - could graduate students ready to tackle any challenges that face them. In the real world few, if any, ideal schools exist. Instead, administrators struggle to offer effective programs. They balance finite numbers of resources, teachers with varied experience and training, varied student/leacher ratios, questionable equipment and facilities, and community representatives who challenge educational decisions.

It is true then, that there must be a distinction made between effective and efficient schools. Though both words are derived from the same Latin root, they represent differing circumstances. The word "effective" suggests that goals are accomplished. Beare et al. (1989) report, "the word, 'efficient', implies productivity, accomplishing an end without waste of effort or resources; it implies getting value for money" (p. 11).

Schools can be effective, but not efficient. Perhaps too much money is spent to gain too little product. Schools can be efficient, yet not effective. The school wastes little, yet the product of the school's endeavours is not able to meet the challenges of society. An appropriate balance must be obtained that maximizes effectiveness while ensuring efficiency. This balance must be recognized when searching for the attributes of an effective school.

Effective School Attributes

The multi-dimansional view of effective schools began to develop in the 1970s. Researchers in the 1980s synthesized research projects to identify specific attributes that typified effective schools. Lists began to emerge. For the purposes of this discussion, three of these lists will be reported.

Ronald Edmonds embarked on a group of studies in the middle to late

21

1970s that led to the identification of five distinct features of effective schools.

These features are tangible and indispensable. Edmonds (1984) described the

features as follows:

- a) Effective schools have strong administrative leadership and have a principal who pays particular attention to instructional quality.
- b) Effective schools have a clear set of achievement expectations below which no student is permitted to fall.
- c) Effective schools possess an "orderly, safe climate" which is conducive to learning.
- d) Effective schools have a well defined instructional focus. School resources and energies are diverted to support that focus.
- Effective schools have established student monitoring systems which ensure that instructional goals are achieved by its students.

Purkey and Smith (1983) reported that the components of effective schools are found in two nested layers established within the structure of a school. Layer two relies upon the strength and the commitment of those responsible for the components in layer one to accomplish improvement goals. The components of the first layer are found in the organization and structure of a school. They can be manipulated readily by administrative and bureaucratic interventions. These include the following components:

- The administration and staff determine the means by which needs are addressed.
- 2. Instructional leaders work to maintain improvement in the

instructional process.

- 3. Staffs keep together and remain stable.
- Especially in the secondary school, curriculum is articulated and organized.
- Schoolwide staff development alters teacher attitudes and behaviours while providing new skills and techniques.
- 6. Parents are involved in and are supportive of the school's work.
- 7. There is schoolwide recognition of academic success.
- 8. Learning time is maximized.
- 9. The school district supports the school.

The second layer of effective schools components includes "process variables" and helps to determine school climate and culture. The components are conceived by those working in the school and are nourished by internal accomplishment and not by bureaucratic manipulation. These second layer components, identified by Purkey and Smith (1983), are:

- 1. The staff engages in collaborative planning and shares collegial relationships.
- 2. There exists a sense of community.
- Commonly shared goals and expectations are clearly stated and guide the decision making processes within the school structure.
- 4. Order and discipline are expected.

Reid, Hopkins, and Holly (1987) compiled an extensive list of study
findings which served as information for educators. The list was organized into 11 specific categories of findings which included: school leadership, school management, school ethos, discipline, teachers and teaching, the curriculum, student learning, reading, pupil care, school buildings, and school size (pp. 24-29). From that list, Reid et al. compiled a 14 point summary of effective school findings:

- The leadership role of the principal and senior management team is critical.
- Schools must be well-managed. A content, well-managed, and united staff is crucial.
- The school is characterized by a favourable school ethos or positive climate.
- The school, both inside and outside classrooms, is orderly at all times.
- The most important factor of successful schools is the quality of the teaching staff.
- The school's energy must be on teaching while promoting empathetic pupil care and learning-centred approaches.
- The curriculum is as important for low as well as high achieving students.
- 8. Students need regular feedback on performance.
- Academic demands must be linked to traditional academic and behavioural values.
- 10. Teachers must reflect high professional standards at all times.
- 11. All classroom teaching time must be used properly.

- Schools must emphasize "traditional" subjects, the basic skills of reading, writing, and mathematics.
- Students should participate in running and organizing their school.
- School structure should not make students feel out-of-place or lost. Students feel more comfortable in buildings which are clean and well cared for.

The lists presented in the literature identified and described traits of effective schools. Themes emerged that emphasized the components of the curriculum and teaching; the impact of an energetic, effective principal whose leadership focuses on classroom instruction; and the dedication of a highly trained staff that assumes responsibility for the successes and failures of student learning. In light of these descriptive lists, critics of the effective schools movement emerged.

The Value of the Effective Schools Movement

Sirotnik (1985) wrote a scathing review of the school effectiveness movement in which he questionned the energy that had been expended to identify common trails of effective schools when educators already knew what worked. The title of his article, <u>School Effectiveness: A Bandwagon in Search</u> <u>of a Tune</u>, adequately summarized a concern that was gaining momentum. In the same year, Sizer (1985) labelled the findings of effective school researchers as common sense. He summarized his views by stating, "A good school does not emerge like a prepackaged frozen dinner stuck for 15 seconds in a radar range; it develops from the slow simmering of carefully blended ingredients." (p. 22). The professors of the effective schools movement were, themselves, relating cautionary messages. Purkey and Smith (1985) stated, "educators should approach effectiveness prescriptions cautiously" (p. 355).

This emerging caution signalled a change and a differing view of the effective school literature. Instead of merely accepting descriptive lists of the conditions that existed in highly effective schools, educators applied these lists to their own situations in an attempt to effect improvement. Critics suggested caution when these lists were used as prescriptions for righting the wrongs of the educational community.

Holmes (1989) states:

The following factors seem to be related to school effectiveness with respect to academic achievement: academic climate of high expectation on the part of teachers and students; universalistic discipline; an orderly atmosphere; frequent and immediate rewards for good performance; regular monitoring of achievement; strong community support; strong leadership. All successful schools do not of course posses all those characteristics. (p. 9)

Holmes continues to express the problem of school effectiveness studies:

It is one thing to discover relationships between certain school variables and certain output variables in existing environments. It is another thing entirely to reproduce such characteristics in other schools, let alone changes in outputs, in very different situations. (p. 10) It became clear that schools needed a "quick fix" for their unique circumstances

and were willing to prescribe effective schools research findings, even though

there were no guarantees that effective schools could be cloned.

There exist many citations in the literature that encouraged school

improvement efforts and acknowledged the value of providing a model for

effective schooling. Davis and Thomas (1989), when discussing the effective

school movement, state:

There is little doubt that a knowledge of research on effective schools and effective teachers can raise awareness of some of what schools and teachers are doing right and what they are doing wrong. Those who are or will be teachers can be guided by this knowledge to improve their own teaching and classroom management habits. (p. 13)

Supporting the value and usefulness of effective schools attribute lists while

acknowledging the limitations of an achievement approach to the research,

Sergiovanni (1991) concludes,

Lists of effectiveness characteristics as proposed by knowledgeable researchers remain useful if viewed as general indicators. They are not so much truth: to be applied uniformly, but understandings that can help principals and others make more informed decisions about what to do and how in improving schools. (0, 91)

Research on effective schooling with its overemphasis on achievement

has carved its niche in the educational literature. Davis and Thomas (1989)

write, "While some consider the effective-school movement to be a passing fad,

most are convinced that the principles are too sound to be ignored" (p.15).

School Improvement

From Planned Change to Educational Reform

Introduction

Clark, Lotto, and Astuto (1984) suggest that school improvement literature can be traced back to the 1930s. Nicholis (1983) suggests that the catalyst for much of the educational change literature can be attributed to the 1957 launching of the Soviet satellite, Sputnik, when the western world perceived that the U.S.S.R. possessed better means to educate and superior levels of technology. No matter what the source, the public has for decades scrutinized systems of education for effectiveness and efficiency.

Reaction to these calls has taken an evolutionary path. Researchers examined the phenomenon of social change with a vision of identifying factors which ultimately led to a positive realignment of the status quo. In the realm of education, researchers embarked on the examination of the educational change process. This, linked with the findings reported in the effective schools literature, led to the deliberate attempt to improve schools - hence the existence of a distinct school improvement literature. The most recent call in the United States has been toward educational reform or restructuring - a call that some believe is an echo that seems to reverberate in waves throughout the educational community over time. Before embarking on a discussion of recent school improvement literature and the more recent issue of educational reform, it is beneficial to revisit earlier descriptions of innovation and planned change. This discussion will serve as a source for essential definitions and basic assumptions that will enable the reader to understand recent trends

Innovation and Planned Change

Innovation and change literature is decades old. It could be argued that the term, change, is now outmoded and reflects a practice in the educational system that was doomed to failure even before it began. However, research examining these failures established a firm footing for present school improvement and educational reform literature.

Jean Ruddick, a professor of education in the United Kingdom, has endured educational change efforts since the 1950s. In a retrospective view, she states:

In our efforts at change I think we have generally underestimated the power of the existing culture of the school and classroom to accommodate, absorb or expel innovations that are at odds with the dominant structures and values that hold habit in place. (Ruddick, 1990, pp. 27-28)

Reflecting on the failed efforts of planned educational change, Ruddick continues:

We should be comforted by the thought that schools are not easily thrown into disarray by curricular fads and fancies, whimsical novelties and light persuasions. The problem is that they seem equally impervious to what we think of as our reasoned, relevant and legitimate proposals for curriculum change. (Ruddick, 1990, p. 28)

In light of these statements, it is beneficial to briefly look back at the literature of planned change.

Planned Social and Educational Change

The social system.

According to Rogers (1962), the social system is defined as a "population of individuals who are functionally differentiated and engaged in collective problem-solving behaviour" (p. 14). Rogers explains that the word, individual, can represent a single person, informal groups, industrial firms, or schools that are linked by a common problem for which they seek a solution.

Change.

Rogers (1962) argues that change occurs as the result of imbalance in the status quo of a social system. Nicholls (1983) sees educational change as "a continuous reappraisal and improvement of existing practice and which can be regarded as part of the normal activity of curriculum development" (p. 4). Leithwood (1986) suggests that the term, "change", is, in itself, neutral. Planned change in a "process of improvement" of 'getting better' in someone's terms" (Leithwood, 1986, p. 2). He argues that educational change "implies at

least a preferred direction if not a valued outcome" (p. 2).

Leithwood (1986) presents planned change as a process by which schools reduce the gap between two distinct boundaries: (a) those things that are valued for students and the objectives schools hope students will achieve, and (b) images of the educated person - "the most succinct and holistic expression of the public's aspirations for their children's intellectual, social, cultural, and emotional develoment [<u>sic</u>]" (p. 2). He argues that prescriptions for change (innovations, new curriculum guidelines, new assessment procedures, etc.) improve school effectiveness only when that gap is reduced.

Innovation.

The realization that a change is necessary to reduce the gap between what is and what should be is the beginning point of the change process. An innovation is deliberately chosen to address the imbalance in the status quo (Rogers, 1962). Nicholls (1983) provides a definition of innovation that is derived from existing definitions in the educational change literature:

An innovation is an idea, object, or practice perceived as new by an individual or individuals, which is intended to bring about improvement in relation to desired objectives, which is fundamental in nature and which is planned and deliberate. (p. 4)

The innovation, it is believed, has to be first adopted and then diffused to the members of the social system.

32

Recent school improvement literature has suggested a different perspective on the notion of innovation. Fullan, Bennett, and Roheiser-Bennett (1990) state, "Innovation should be seen as points of departure or catalysts, rather than as things to 1 implemented" (p. 14).

Guskey (1989) recognizes the number of innovations available for planned change efforts. "At no other time in the history of education have there been more new ideas and innovations available to educators" (p. 11).

Adoption.

Adoption occurs when members of a social system decide to test an innovation. Adoption implies that individuals or groups of individuals have weighed the potential consequences of implementation and are satisfied with the power and/or the improvement that their deliberations espouse. Havelock (1973) suggests that another name for adoption is acceptance. Adoption has occurred when members of a social system have accepted an innovation.

Diffusion.

Diffusion, according to Rogers (1962), is the process by which an innovation spreads from the source of conception to those who will ultimately use it.

Intervention.

Hall (1988) argues that the term, intervention, has been a crucial concept in the educational change literature, yet it has rarely been defined. In response, he develops a definition of the concept for use in his own research: "An intervention is an action or event, or set of actions or events, that inluences the use of an innovation" (p. 51). Hall points out that references to who made the intervention, whether there are observable effects, and when it occurs have been intentionally left out of the definition. These aspects, despite their importance for intervention description, are not critical to defining the concept.

The change agent.

The existence of a person responsible for facilitating change in a social system can substantially increase its rate. Authors refer to this person as the change agent (see for example Rogers, 1962; Schaller, 1972; Havelock, 1973). The change agent may be chosen from inside or outside the social system depending upon the circumstances delineating the need. Havelock suggests four primary ways in which a person can act as a change agent:

- 1. by acting as a catalyst for change;
- 2. by providing solutions to problems;
- 3. by helping the process of change move along; and

 by linking resources for those involved in the change effort.
In education, change agents can be district administrators, school administrators, or teacher leaders.

The change facilitating team.

Studies from the early 1980s suggest that one of the reasons that educational change failed was that the administrator did not have enough time to facilitate and manage the change. Fullan (1984) reports that "a large percentage of principals (at least one-half) were preoccupied with administrative work and organizational maintenance activities" (p. 100). To address this, it was suggested that the principal play a facilitative leadership role on a change facilitating team (see for example Hall, 1988). The principal would work with other change facilitators and would orchestrate the change effort in a shared, cooperative manner. The traditional role of change agent was reassigned to a change facilitating team.

The change facilitating team, Hall (1988) suggests, typically consists of four members. The principal is regarded as the primary change facilitator. A colleague such as the vice-principal, department chairperson, or teacher with specific specialist knowledge plays a supporting role as the second change facilitator. A third level of facilitator, the teacher, plays a role that is less formalized. Members of this third group model new practices for colleagues. disseminate information to other teachers, cheerlead, and provide support.

Sparks (1993) recognizes the value of a team approach. One of his 13 tips for managing change in schools suggests that a team approach ensures that all stakeholders have an essential role in the improvement process. He states, "The synergy that flows from people with various perspectives produces higher quality ideas and better solutions" (p. 14).

The change facilitating team is led by the school principal. Sergiovanni (1991) states, "The principals who were most effective in implementing change were team-oriented, working closely with these other levels of change facilitators" (p. 268).

Earlier literature.

One thrust of the planned change literature is to describe the circumstances that lead people to adopt an innovation. Rogers (1962) proposes a five stage decision making process through which each individual passes for a change to occur. These five stages are: awareness, interest, evaluation, trial, and adoption.

Another thrust of the planned change literature is to describe the climate which ultimately leads to change. Schaller (1972) identifies 12 characteristics of a "creative organization" that possesses a climate conducive to change. The creative organization must have a contemporary orientation; be aware that problems exist; tocus on people or their needs rather than product; emphasize problem-solving; be aware of the importance, relevance, and availability of knowledge; monitor the pace of change; evaluate its present operation against its perceived purpose; realize that the organization will frequently attempt to reach a goal; place the emphasis of financial administration on output rather than income; be committed to maximizing the problem-solving capability of the personnel; have leaders open to new ideas; and allow personnel to experience all aspects of the system.

The literature reflects the realization that the pace of change is exasperatingly slow (see for example Schaller, 1972, p. 64). Rogers (1962) refers to the time between the insight of a problem and the introduction of an innovation as a 'time-lag'. It was observed that in the face of social or technological crisis, the speed of change increases substantially (see for example Schaller, 1972). There is a belief that if one could cause an imbalance in the status quo by intervention and manipulation, then the process of change would proceed more rapidly (see for example Havelock, 1973). Later studies describe social climates that ultimately increase the rate of adoption (see for example Schaller, 1972).

Many of the failures cited in early literature are attributed to the resisters of an innovation. A resister is defined as a member of the social system who hesitates to adopt an innovation because of a perception that change is not

37

advantageous to the social system or is a threat to its actual being. Nicholls (1983) reflects on the resister's role in planned change. It is through resistance that a social system becomes aware of the consequences of the implementation of an innovation and its impact on the entire efficiency and stability of a social system. The innovator must "win' the support of the resistor(s) by thinking through the reasons for resistance and by embarking on an effective campaign to make the advantages and the positive effects known.

The Process of Planned Educational Change

Fullan (1982) states, "Change is a process, not an event" (p. 42). In his discussion of this complex process, he identifies three phases of educational change reported in the change literature: <u>Phase I</u>: Iniliation; <u>Phase II</u>: Implementation; and <u>Phase II</u>: Continuation. These phases are entangled and rarely can be delineated one from another.

Fullan (1982) suggests that entwined in each phase of the change process, there exist numerous factors acting and interacting. One cannot effectively separate them into distinct entities. Instead, change must be viewed as an interactive process where decisions made at one point of the continuum can feed back to alter decisions made at an earlier stage. The opposite is also true. That which occurs at an earlier stage strongly affects that which happens at subsequent stages.

Phase I: Initiation.

The change literature uses a variety of terms to describe the forces that begin the process of change. These terms include initiation, mobilization, and adoption. In this phase, the members of an educational system become aware of an innovation, grasp it, and decide to proceed with implementation.

The pervasive view that innovations are adopted because they are desirable may not be true to all change situations. Instead, Fullan (1982) contends that there are a myriad of factors associated with the decision to adopt an innovation. Ho identifies a ten point summary list:

- 1. Existence and quality of innovations
- 2. Access to information
- 3. Advocacy from central administrators
- 4. Teacher pressure/support
- 5. Consultants and change agents
- 6. Community pressure/support/apathy/opposition
- 7. Availability of federal and other funds
- 8. New central legislation or policy (federal/state/ provincial)
- 9. Problem-solving incentives for adoption
- 10. Bureaucratic incentives for adoption
- (p. 42)

The presence or absence of each factor or combinations of factors can

influence the decision to adopt or reject an innovation.

Phase II: Implementation.

The terms related to the second phase of the educational change

process include implementation and initial use (Fullan, 1982). Fullan states:

Implementation consists of the process of putting into practice an idea, program, or set of activities new to the people attempting or expected to change. The change may be externally imposed or voluntarily sought; explicitly defined in detail in advance or developed and adapted incrementally through use; designed to be used uniformly or deliberately planned so that users can make modifications according to their perceptions of the needs of the situation. (p. 54)

In this phase, the members of an educational system decide to use the

innovation in an attempt to assess its value and power. Fullan contends that

because more people are involved in this phase of the process and because

successful change is at stake, this phase is intricate and roust be handled

delicately. What people do and do not do can ultimately end in the rejection of

the innovation.

Fullan (1982) organizes the major factors affecting implementation into four distinct categories:

 <u>Characteristics of the Change</u>: Implementation will occur more readily if members of a social system perceive that there is a need for the innovation; if the innovation is clear about what teachers should do differently; if complex innovations have been broken into small, clear incremental components; and if the materials supplied for the implementation process are practical and well suited to the needs of the teachers.

- 2. <u>Characteristics at the School District Level</u>: Implementation will occur more readily if the district has in the past experienced successful change; if the process of adoption reflects careful consideration, planning, commitment, and follow-through; if there is support by district administrators; if in-service and preparation time has been provided for staff development; if a realistic time-line and a system of evaluation have been established; and if the school district has effectively prepared and gained the support of the oublic before introducing an innovation.
- 3. <u>Characteristics at the School Level</u>: Implementation will occur more readily if the principal actively supports the innovation; if teacher relationships are characterized by supportiveness, openness, collegiality, trust, and helpfulness; and if teachers' own sense of efficacy (whether teachers think and expect that all children, regardless of social background, can reach appropriate levels of achievement) is strong.

4. <u>Characteristics External to the Local System</u>: The external environment provides influences that aid in implementation. If government agencies support the implementation of an innovation and if appropriate external assistance such as technical expertise is available, then implementation is more likely.

In conclusion, there are numerous factors associated with successful implementation. With widespread support, ample professional preparation, detailed planning for implementation and continuation, and access to appropriate materials and expertise, it is likely that an innovation will be implemented successfully.

Phase III: Continuation.

The third phase of successful educational change has been referred to as continuation, incorporation, routinization, or institutionalization (Fullan, 1982). It is in this phase that an innovation becomes a "built-in" part of the daily practices or attitudes of a social system, or is discontinued and allowed to die a quiet, peaceful death.

The importance of the relationships shared by teachers and administrators cannot be overestimated at this stage in the process. Miles (1983) reflects on his experiences with the <u>Study of Dissemination Efforts</u> Supporting School Improvement, hereafter referred to as the DESSI study:

Administrators push, demand, support, and think about the organization; teachers react, get involved, struggle with the demands of the innovation, and think about their lives with students. It was very clear that an underlying variable we called teacher-administrator harmony was critical for success. (b. 19)

Harmonious interrelationships among the players in the educational process are crucial for continuation and institutionalization.

Miles (1983) suggests that an innovation must become a part of the internal structures and functions of a school otherwise reversion to past beliefs and practices may result. He states, "Without some sense of 'built-in-ness' the fate of innovations is in doubt" (p. 14).

Fullan (1982) makes two important points about the discontinuance of implemented innovations. First, he suggests that in the period following effective implementation, innovations can be discontinued when external agency funding ends. If this happens, schools could face an extra financial burden that cannot be addressed by reallocating existing funds. The only alternative , to discontinue the use of the innovation. Secondly, Fullan suggests that changes in school staff can also cause discontinuance. New teachers and administrators may not perceive the need to use the innovation or may not possess the expertise or commitment that former staff members did.

In conclusion, the ultimate goal of change is to make an innovation a regular, built-in part of the daily life of a school or school district. Continuance

43

or institutionalization can be affected by various factors. Although careful planning can counter discontinuance, unforeseen factors can have a devastating effect on the institutionalization of an innovation.

Summary.

Change is an evolutionary process. In this process, there exist three phases: Initiation; Implementation; and Continuance. These phases do not have distinctly delineated boundaries. Actions related to one phase affect the processes in another. The interacting forces determine the ultimate fate of an innovation. People involved in planned change must accept that what they do or do not do can determine the success or failure of a planned change venture.

The Players in the Process and their Roles

In any educational change endeavour, people are involved. Researchers have studied the roles that people have played in successful and unsuccessful change efforts and have reported on the actions that lead to implementation.

District level administrators.

The support of district level administrators and their subordinates determine the success of districtwide change. Fullan (1982) slates, "What they [district administrators] do at each of the three main phases of the change - the initial decision or mobilization, implementation, and institutionalization significantly affects the destiny of the proposed change" (p. 163). The literature identifies characteristics of district administrators that significantly support each phase of the change process.

During Phase I, the district administrator sets the tone of impending change. Teachers perceive that if the superintendent shows a convincing commitment to the innovation, then it must be important, necessary, and should be taken seriously. This commitment must make more of an impact than simple verbal or general support. Instead, funds must be allocated to purchase appropriate resources and opportunities must be provided for profossional staff development. If necessary, professionals with expertise in the innovation must be brought to the district to provide guidance and training for those cirectly involved in the change process.

Fullan (1982) suggests that the participation of the district administrator in Phase II is more important than the commitment demonstrated in Phase I. He suggests that the way in which the implementation plan is construed will determine the ultimate institutionalization or continuation of the innovation. District administrators must allow the administrative power to be shifted from the central office to the school so that those directly involved in implementation are given the opportunity to adapt, redevelop, and reflect on the innovation in relation to their unique circumstances. The district administrator must provide a means by which teachers can access technical information about the

innovation, address problems encountered in the implementation, and

communicate successes and failures to colleagues in the process.

Riffle (1987), while reflecting on school improvement initiatives,

discusses the empowerment of schools. For this empowerment to occur, the

district administrator must relinquish some power so that schools can develop a

commitment to the change. He states:

The developmental argument is that the empowerment of schools must be seen as a condition for their improvement; people will not undertake development unless they believe that it will be fruitlut, that it will give them greater control over their work, and it will increase the likelihood of success. Giving people in achools the autonomy to choose directions, but not the freedom to do nothing, may be the best way to combine a concern for the improvement of all schools in a system with both respect for the professionalism of educators and a recognisan that the opportunity for self-direction can be a powerful stimulus to development. (p. 3)

In Phase III, the district must continue to provide funds to ensure that the

new needs of the district are met. How the district administrator plans for

continuation is crucial. The method by and ease with which the new program is

incorporated in district budgets contributes significantly to the continuation of

the innovation.

School principals.

The school principal is ultimately responsible for preparing a school for

change and for directing the change process. The principal's behaviour can affect the outcome of a school improvement effort. Clark et al. (1984) suggest that principals influence the members of a social system through "suasion and the assertion of high expectations" (p. 54). They report that the actions of effective principals include communicating the innovation's importance and implementation success, arranging for training and materials, and rescheduling time for practice and reflection. Sergiovanni (1991), reflecting on school improvement, states:

School improvement does not occur by happenstance. Someone must decide to do something to change the status quo for the better. Sometimes the decision to embark on school improvement efforts emerges from a teacher or a group of teachers; generally, though, such efforts result from deliberate action by the school principal. (p. 255)

Hall (1988) suggests that the school principal should be viewed as the leader of the change facilitating team. Through frequent discussions, this team facilitates the change process by modelling the innovation, disseminating pertinent information to teachers, providing support, anticipating the results of specific actions, and troubleshooting where necessary. The more effective leaders are team-oriented and can work closely with other leaders of a change facilitating team.

Hall, Rutherford, Hord, and Huling (1984) describe three change facilitator styles: Initiators, Managers, and Responders. The following 47

discussion provides a summary description of these principals.

- Initiators Initiators have a vision of what good schools and teaching should be like. They have clear, decisive long-range policies and goals that are based on current educational research. Initiators work intensively to implement these policies and goals. When changes in district programs and policies are sought, input from staff is solicited. Decisions are based on the goals of the school.
- 2 <u>Managers</u> Managers exhibit a broad range of behaviours. Their reactions are linked to their rapport with teachers and central office stat!. They are sensitive to teacher needs and work without fanfare to support teachers as they use an innovation. Managers do not generally initiate attempts to move beyond that which has been imposed.
- 3. <u>Responders</u> Responders allow teachers and others to take the lead. Their focus is on administrative tasks, keeping teachers content, and treating students well. Responders view teachers as strong professionals who are able to carry out their instructional roles with little guidance. They provide everyone with an opportunity to voice an opinion before making a decision; sometimes decision-making is passed to

others. Responders have a limited vision of the future of their schools and their staf's and tend to make decisions based on immediate circumstances rather than on long-range goals. They try to please others.

Hall et al. (1984) report that initiator principals are most associated with successful change efforts. Responder principals are least likely to be associated with successful change endeavours. Sergiovanni (1991) states.

Initiators have a clear sense of what noeds to be accomplished and take more active roles in planning, prodding, encouraging, advising, participating, checking, stimulating, monitoring, and evaluating change efforts. Further, they assume more direct roles in obtaining and providing the necessary material and psychological support for successful change efforts. (p. 266)

Teachers.

Fullan (1982) summarizes the teacher's role in the educational change process: "Educational change depends upon what teachers do and think - it's as simple and complex as that" (p. 107). Innovations must work for teachers as well as for students. This point can be easily overlooked for school improvement efforts are evaluated primarily on improved student outcomes. If an innovation does not fit into the daily work of a teacher, then the probability of institutionalization is small (Clark et al., 1984). Early research reports that in order for change to be successful, teachers have to be involved in the initial phase of the change process. They have to develop a personal commitment for the innovation. Whether by active intervention on the part of a change agent, or by some other powerful means, teachers have to believe that change is beneficial and have to be involved in choosing the appropriate initovation. Recent studies have found that the administrator, not so much the teacher, must believe in and adopt the innovation. Teachers develop an appreciation for the innovation as it is being implemented. Guskey (1987) observes, "The most significant changes in teacher attitudes and beliefs come after they begin to use a new practice successfully and see changes in student learning" (p. 57).

Sparks (1993) proposes that some teachers act as "paradigm shilters" and should be sought out at the beginning of a change effort. He suggests, "Paradigm shifters create innovative approaches to solving intractable problems long before others see the need for a new set of rules and procedures" (p. 15). These teachers can experience the innovations first and can provide collegial support as others embark on their part of a school's mission.

There have been inconsistent literature claims about teacher characteristics of and the probability of successful school improvement efforts. One trait, however, has emerged - the teacher's sense of efficacy. If teachers feel that they can be effective in teaching students and can effectively change,

50

then the chance of school improvement success is great. Davis and Thomas

(1989) state,

Apart from any district-wide or school-wide formal improvement program, an individual teacher who wishes to improve his or her skills must be highly motivated (and sufficiently secure) to engage in self-examination and self-improvement. (p. 182)

Crandall (1983), reflecting on his involvement with the DESSI study, lists

factors that contribute to the success of teachers engaged in improving

classroom practice: commitment, exemplary practices, training, and

administrator leadership. He comments on the interaction among these factors:

We found these factors interacting in some quite uncommon ways that resulted in, among other things, highly committed teachers using with a high degree of lidelity innovations they had neither developed or [sic] chosen. (p. 17)

The message is clear. Teachers do not have to choose an innovation.

Instead, they must play a role in planning for and making decisions about the

implementation, not the adoption, of an innovation. It is through this process,

with continuous support and assistance, that commitment and or nership are

developed.

Internal assisters.

Internal assisters are school district personnel who have expertise in specific areas of curriculum or in grade-level programming. The planned change literature labels these specialists differently. Fullan (1982) refers to them as internal district consultants; Cox (1983) writes about the central office staff; the Newfoundiand Departm⁻nt of Education reters to them as program coordinators, and Clark et al. (1984) fail to mention them or their contributions at all. By whatever name, internal assisters "can perform critical functions that make school improvement really work" (Cox, 1983, p. 13).

Fullan (1982) warns about drawing broad-based conclusions regarding the role of internal assisters. He argues that few research projects exist that have as their focus the internal assister. Generalizing from a small number of case studies and surveys, he suggests, is dangerous.

To understhind the internal assister's role in planned change, it is valuable to revisit the conclusions of the DESSI study. The role of the internal assister was enamined by Cox, the result of which was the identification of a nine point list of contributions that aids in the institutionalization of an innovation. Cox (1983) found that central office staff (internal assisters):

- Became familiar with the needs of students in individual schools in their districts.
- 2. Located and helped select the new practices.
- Knew the content of the new practice, its purpose, and the benefits that were to result from its use.
- Helped arrange and conduct training in the new practice, working with external assisters.
- Arranged funding and other support from the district or other sources.

52

- Obtained endorsements for the practice from the superintendent, school board, principal, and teachers.
- Worked with the teachers using the classroom, working out 'bugs' and overcoming obstacles.
- 8. Assisted in evaluation.
- Helped plan how to continue and institutionalize new practices. (p. 12)

Cox states,

Their (internal assisters) help contributed to more implementation outcomes than any other single group of assisters, having impact on the amount of change in teachers' practice, the level of mastery teachers achieved, the commitment teachers developed, and the benefits they perceived from using the practice. (b. 12)

External assisters.

Fullan (1982) reports that although few studies have examined the contributions of external assisters to enduring change, those that do exist place external assisters toward the bottom of the list of influential resources (p. 184). In his discussion, he names the limited number of external assisters available to support Canadian educational change endeavours. These assisters include government ministry of education personnel whose primary

responsibility is policy and program dissemination and effectiveness research;

members of a university faculty involved in professional development research;

and teacher unions or associations.

Clark et al. (1984) suggest that despite conflicting research evidence, the

role and the value of external assisters in the school improvement process is

valuable. They state,

Because of the resocialization needs of the change process, external assisters must have continuous contact with the school-level implementers. To be effective at the school level, the assistance offered must be personal and practical. In addition to technical expertise, external facilitators must be knowledgeable of the change process and be sensitive to the critical role played by building and district level administrators. (p. 55)

Cox (1983), reflecting on the DESSI study, compiled an 11 point list of

external assister contributions found to be the most effective and most valuable for

school improvement success. External assisters:

- 1. Made school people aware of the existence of new practices.
- Helped school people choose among a range of new practices, matching local needs with an adequate resource.
- 3. Sometimes helped arrange funding for the new practice.
- Worked with local administrators, teachers, and school boards to develop commitment to the new practice and arrange for it to be installed.
- 5. Arranged and conducted training in how to do the new practice.
- 6. Worked with a local content person on the new practice.
- 7. Provided new materials for the new practice.
- Worked through the details of the new practice with teachers, planning implementation schedules, and paying attention to the specifics of actually using the practice in the classroom.

- 9. Evaluated new practice and analyzed data.
- 10. Provided follow-up help as implementation progressed.
- Helped develop plans for continuation and institutionalization, for example, securing additional funds and developing new users at school. (pp. 11-12)

Cox suggests that "of these activities, the most helpful for teachers were the

efforts to actually work through the specifics of using the practice in the

classroom" (p. 12).

In summary, Cox (1983) states:

their [external assisters] major contribution to school improvement appears to be preparing a congenial environment for the new practice (ensuring that resources, facilities, and so forth are in place), rather than assisting with the content of the new practice. (p. 12)

School board and community.

Clark et al. (1984) argue that school boards, parents, and communities

do not have a distinct role in the decision to implement innovations (p. 55).

Fullan (1982) states:

by far the most prevalent case is that school boards and communities do not initiate or have any major role in deciding about innovative programs; that is, administrators and teachers develop or make recommendations about most new programs, or governments legislate new policies. (p. 194)

The school board and the community play a role in the realization that change

is necessary and in the rejection of innovations.

Fullan (1982) states that school boards and communities "can be radically powerful in the smaller number of cases where, for whatever reasons, they become aroused" (p. 194). One of the more prevalent reasons for becoming aroused, he suggests, is a shift in population. To support his contention, he provides an example of a community that underwent a demographic change. The stagnant school system was targeted by the new members of the community; new educators were hired; a new school board was elected; and positive changes were implemented. He warns that these kinds of community initiatives that result in positive change are relatively few. More often, these population changes can cause enduring conflict.

A second influence of school boards and communities on planned change is that they can, if members have been ignored in the change process and if members possess the appropriate resources, bring about the end of an adopted innovation. Fullan (1982) suggests that these endeavours reflect process failure. The board and the community act as gate-keepers to bar illconceived innovations from schools. The price of such rejections can be failed and abandoned innovation, poor morale, and attrition of administrations and teachers (p. 195).

According to the literature reviewed on the change process, school boards and communities play a minuscule role in planned change. Generally, they do not actively participate in the adoption and implementation processes,

56

but can effectively prevent change when members are dissatisfied with the results of implementation.

Converging Literatures

The school improvement literature has a purpose that is different from all other educational inquiry. It explores the ability of a school or school system to adopt an innovation and reduce the gap between what is and what should be. It measures the level of adoption while assessing the means by which the adoption occurs. It examines the impact of leadership on a school's ability to change and reflect contemporary expectations while describing the roles of other key players in the pursuit of improvement.

Clark et al. (1984) argue that school effectiveness research and school improvement research share input and output variables. They support this contention by citing common themes found in both literatures. "They have examined leadership, school climate, teachers, students, curricular materials, patterns of curricular organization, instructional tactics and strategies, financial resources, facilities and equipment, and parental and school-community involvement in education" (Clark et al., 1984, p. 42). School improvement researchers examine the role of the principal in institutionalizing an innovation.

As the early literatures of educational change and effective schools evolved, they followed parallel paths. Toward the late 1970s and early 1980s, these paths began to converge. The characteristics of effective schools, as reported in the effective schools literature, were being construed as innovations. Many administrators saw these lists as quick fixes for their ineffective schools. If an administrator perceived that the academic achievement of students needed improvement, then implementing the characteristics reported in the effective school literature was a means for success. The school improvement literature provided guidance and a focus for implementing innovations. Later, as models of school improvement emerged, the school improvement literature was used to guide the implementation of other innovations.

School Improvement

Susan Loucks-Horsley and Lesley F. Hergert (1985) suggest a broad meaning for the term, "school improvement". They state:

We use this term quite broadly to include the pursuit of any goal that benefits students and that has as its focus the classroom and school building. These goals may include, as have our own experiences, the following:

 Changing an elementary school's reading program in response to low achievement scores.

 Developing and implementing a districtwide science curriculum for elementary grades.

Implementing Edmond's "correlates of effective schools" in high schools.

Applying advances in technology to a district's special education

program.

 Increasing equitable treatment and opportunities for all students in a large urban school district. (p. viii)

In the preface of Huberman and Miles's 1984 publication entitled,

Innovation Up Close, there exists a paragraph that adequately states the

challenge of true school improvement efforts:

School improvement, like motherhood, has many advocates. Everyone is to it, without having to campaign actively on its behalf. And just as the 100% of people who have had mothers think they know how mothering could be done 'wetter, so the (nearly) 100% of people who have been pupils in schools, or have even taught in or managed them, think they know how schools can be improved. More procisely, they are sure that schools *ought* to be improved. The trouble is that they propose a staggering, conflicting range of methods of improving schools, from back to the woodshed' to teacher meri pay, a stilfer curriculum, a stronger tax base, reorganization, a more humane climate, teacher-proof innovations, community involvement - the list is nearly endless. (p. v)

Huberman and Miles continue the discussion and highlight the problematic

nature of the word improvement. "Improvement sometimes turns out to be a

code word for the directives that administrators have successfully put into place,

or for the agreements that teachers have lobbied into being" (p. v).

The theme of school improvement.

Clark et al. (1984) summarize the pervasive theme of the school

improvement literature:
The basic message to be derived from the research on school improvement is as unambiguous and optimistic as that of the school effectiveness research:

- 1. Public schools and school systems can and do improve;
- 2. School improvement programs (federal, state, and local) work; and
- Professional educationalists are capable of effecting positive educational changes. (p. 50)

This optimistic view, however, has not always existed in the minds of all

educational change researchers. Philip Runkel (1984) candidly shares his view

of educational change:

I worry about the fact that the more things change, the more they stay the same. It is also true that the most of those things soon fade away, and things go back to normal. It has almost come to the point where, when we speak of an innovative school, we mean one that tries one new thing after another without making any of them work. (p. 178)

Educational researchers continue to study the forces that ensure enduring

change.

Summary of school improvement research.

Clark et al. (1984) summarize the school improvement research findings

by presenting a set of propositions.

<u>Proposition 1</u> - Public schools, individual classrooms, and school systems can and do improve, and the factors facilitating school improvement are neither so exotic, unusual, or expensive that they are beyond the grasp of ex::aordinary leaders in ordinary situations. Proposition 2 - People matter most in school improvement programs:

- (a) Teachers can and will implement new practices and programs given active leadership from building and central office administrators, a chance for planning the implementation process, appropriate training, opportunities for interaction, breathing space to try and fail, and continuous assistance and support;
- (b) Building level administrators make a difference in school improvement programs by establishing a climate of expectations that teachers will successfully improve practice and by providing on-site coordination, communication, assistance, and s:pport;
- (c) District level administrators affect school improvement programs by exhibiting active backing in the form of communicated expectations for success, psychological support, needed resources, and local facilitation assistance; and
- (d) External assisters are most effective at the school level providing concrete and practical assistance on implementation issues, such as planning, scheduling, problem solving, and follow through.

Proposition 3 - An innovation is more likely to be adopted and implemented if it is perceived as having relative advantage, compatibility, simplicity, and legitimacy. Implementation is more effective when the innovation tocuses on a specific need and demonstrates catrily in purpose and techniques.

Proposition 4 - Specific resources are necessary to support effective school improvement programs:

- Staff development programs that are task-specific and provide on-going, continuous assistance and support; and
- (b) Monetary resources that are adequate to provide the

people, materials, and time needed in the program. (p. 59)

The set of propositions offered by Clark and his colleagues are a synthesis of the planned change and school improvement literature discussed earlier in this chapter.

A theme has emerged from these literatures. In any successful school improvement venture, the principal is the prime facilitator, teachers are the key players, and all others expend their energies and resources to ensure that the venture is successful.

Models.

Models of school improvement have been highlighted in the literature. Sergiovanni (1991) reports that the systems view of change has driven the conceptualization of specific models. In this systems view, four individual units of change interact and all require attention:

 <u>The Individual Teacher</u> - In any change endeavour teachers count. Though most teachers usually work in isolation, in a school improvement endeavour, all teachers must be included. They are the ones who directly influence the student in the classroom.

- <u>The School</u> School culture is the unique sense of order and being that emerges from norms, customs, and traditions existing in a school. It defines acceptable behavioural limits, beliefs, and worth. Culture is a powerful factor in fostering school improvement. Improvements that are consistent with a school's value system are more likely to be accepted.
- 3. <u>The Workflow of Teaching and Schooling</u> Principals must place a great amount of effort in developing the commitment of its staff to improve schools. Making the specifics of the improvement known to teachers; communicating how the improvement will manifest itself; and providing the necessary support resources, evaluation, supervision, and in-service training will ensure that the improvement attempt is successful.
- 4. <u>The Political System</u> For institutionalization of an improvement, the superintendent, school board, and teachar's association must embrace it. Manipulation and intervention on the part of the principal may be necessary to communicate the successes of an improved practice.

It is suggested that when each unit of change in the system has been addressed appropriately, school improvement will result. Sparks (1993) suggests, as one of his 13 tips for managing change in schools, that those planning a change endeavour should use a systems approach. Changes will not occur when the innovation touches a fragment of the system.

Many models have emerged that are based on this systems view of change. Beare et al. (1989) document the Caldwell and Spinks school management model entitled, <u>The Collaborative Management Cycle</u> which incorporates school management and change with policy writing and implementation. In 1978, Hall and Loucks proposed <u>The Concerns-Based</u> <u>Adoption Model</u> that Loucks-Horsley and Hergert (1985) authored and the Association for Supervision and Curriculum Development published as <u>An</u> <u>Action Guide to School Improvement</u>. To understand the thrust of a school improvement model, the author will briefly highlight the latter.

The concerns-based adoption model.

Teachers and administrators involved in school improvement initiatives express concerns about the innovation. In 1976, Hall and Loucks proposed a continuum of seven hierarchical stages of concern through which teachers and administrators pass when engaging in the process of school improvement. The first set of concerns are self-oriented: What is the new practice? How will the innovation affect me? These soon give way to management concerns: How do I do it? How can I keep it from taking too much time? When the management tasks are mastered, the final set have to do with the impact of the new practice(s) on the student: Are they learning? What can I do better? This continuum aided in the development of <u>The Concerns-Based Adoption Model</u> (hereafter referred to as CBAM).

CBAM is a model of school improvement that is based on a seven step linear progression. Loucks-Horslev and Hergert (1985) argue that

the linear process imposes some order on a complex situation. However, if you pick up the book in the middle of your project, you may find that you have already done some of the early steps instinctively. And even if you start at the beginning, you may find yourself working on two steps at once. (p. xii)

The following discussion is a brief overview of the model.

Step 1: Establishing the School Improvement Project - In implementing

the first step of the model, schools are encouraged to clarify their visions,

negotiate for resources, build relationships and identify allies, consider using

outside consultants, and form a school improvement team.

Step 2: Assessment and Goal Setting - Although it is recommended that school improvement teams should not spend a great amount of time assessing existing school circumstances, there must be a sense of where the school is at the start. The school improvement team must define the problem, collect data, analyze the data, and determine a shared vision of the improved school. Then, goals for the school improvement effort should be set.

Step 3: Identifying an Ideal Solution - Keeping the present school circumstances in view, the school improvement team embarks on a quest to find the ideal solution. The team identifies local resources and constraints, develops criteria for the solution (while recognizing that these criteria may change as the project unfolds), locates outside resources, applies criteria for solutions, makes a decision about the actual solution, and transforms the solution into a definable practice.

<u>Step 4: Preparing for Implementation</u> - At this point in the model, the school improvement team prepares to implement the solution. This preparation includes seven activities: creating awareness, selecting implementors, assessing current practice, setting expectations, assigning support roles, making logistical arrangements, and creating a timeline of activities and events.

Step 5: Implementing the Project - Here, the focus shifts from the school improvement learn to those implementing the solution. The management of this stage involves continued training, and providing on-going support.

Step. 6: Reviewing Progress and Problems - Following the early stages of implementation, the school improvement team must assess the success of the implementation. The members must analyze the perceptions and the progress of those involved in the implementation, evaluate the initial outcomes of the implementation, and make appropriate refinements.

<u>Step 7: Maintenance and Institutionalization</u> - Planning for maintenance and institutionalization should be a continual goal in all steps of this model. Specifically, however, the school improvement team must ensure that all

administrators continue to support the improvement and all staff members renew their commitment and their skills.

As can be ascertained, CBAM embraces the planned change literature and provides schools with a conceptual framework for school improvement.

Reform: A Call for Restructuring the Education System

There has been a recent cry in the United States for educational reform. In Canada, the cry has not been as strong, yet Canadian educators are again beginning to listen to neighbours involved in restructuring schools. Since the initial cries, "a more recent wave of reform initiatives - commonly labeled under the rubric of restructuring - calls not for repair of the current system but for a reshaping of the entire educational enterprise" (Hallinger, Murphy. & Hausman, 1992, p. 330).

Hallinger et al. (1992) explain the thrust or new wave of restructuring schools:

Basically, restructuring includes endeavours to (a) decentralize the organization, management, and governance of schooling; (b) empower those closest to the student in the classroom (i.e., teachers, parents, and principals); (c) create new roles and responsibilities for all the players in the system; and (d) transform the learning-teaching process that uniolds in classrooms. (p. 330)

Fullan and Miles (1992) suggest that this new wave of reform comes as the result of planned change failures. Earlier attempts at making schools better, failed because of the lack of cohesive plans based on solid educational change knowledge. Instead, Fullan and Miles suggest, "Rather than develop a new strategy for each new wave of reform, we must use basic knowledge about the do's and don'ts of bringing about *continuous improvement* (p. 745).

Joyce (1991) groues that rhetoric attached to the call for restructuring leads him to believe that restructuring is the current name for school improvement. The call for collegiality within a faculty, sharing site-specific information such as student achievement with the faculty, and providing curriculum initiatives are echoes from the school improvement movement.

Summary

The school system has been involved in educational change for decades. Historically, there have been waves of interest in changing the educational system. These waves have been named educational change, school improvement, and restructuring. Each has been built upon the findings of the previous wave making its thrust more powerful and ensuring more success.

The message of the evolving literature is clear. Change (or improvement, or rectructuring) is a slow, arduous process that requires a clear vision, knowledge of the change process, knowledge of a proposed innovation, careful planning, exceptional leadership, constant support, unfaltering insight,

calculated risk-taking, and common sense. As Riffle (1987) suggests, mistakes will be made. People are involved. However, through commitment, practice, and feedback change will become a normal part of a dynamic school system.

CHAPTER THREE METHODOLOGY Research Design

According to Yin (1984), "a research design is an action plan for getting from here to there, where 'here' may be defined as the initial set of questions to be answered and 'here' is some set of conclusions (answers) about these questions" (p. 28).

Research Phases

This research was completed in two distinct periods using two datagathering techniques. Each phase was characterized by a specific research question. Data were gathered in the first phase using a survey instrument. In the second phase, data were gathered by conducting a case study of four of the 23 survey respondents.

The First Research Question

The first research question, intended to be of an exploratory nature, is stated below: <u>Research Question #1</u>: According to individual school principals, what factors contributed to the increase in the mean grade six composite scores on the <u>Canadian Tests of Basic Skills</u> in their respective schools during the period between 1985 and 1991?

The intent of this question was to isolate school principals' perceptions of factors that had contributed to the increase in composite scores over the vix-year period. To gather these data, Yin (1984) suggests that the researcher use a quantitative research tool, the survey. According to Yin, research questions that seek to find answers to "what" questions, can effectively be answered using a survey or questionnaire strategy (p. 17).

The Survey

Development of the Instrument

The survey was designed in April 1992 following an extensive search of effective school literature. It was found that many authors cite J. S. Coleman's 1966 report to the congress of the United States of America entitled, <u>Equality of</u> <u>Educational Opportunity</u>, as the catalyst for much of the effective schools research (for example see Robinson, 1985; Davis & Thomas, 1989). Robinson reported the Coleman Report finding that factors that could not be manipulated within the structure of the school determined the achievement level of its students. In later years, researchers acknowledged the power of socioeconomic status in determining achievement. They did, however, feel that other factors that could be manipulated within the school setting played a valuable role in determining student achievement levels.

Because of these literature findings, the survey contained live sections. The first two were entitled <u>Community Changes</u> and <u>Curricular Changes</u>. <u>Community Changes</u> included statements about changes in the school's population, the community's socioeconomic situation, and the community's population. <u>Curricular Changes</u> contained statements that explored recent curriculum initiatives, innovations, staffing changes, and other factors that could have contributed to the increased composite scores.

A third section entitled, <u>Community Factors</u>, was included to address another burgeoning area of effective school research. It is believed that the perceptions and participation of the public can affect the achievement levels of students. The statements in this section of the survey explore the principal's perceptions of the community's support for ...e school, the school system, and the school's programs.

To provide a framework for the data collected and to link the respondent's school to the initial CTBS data, a demography section was included in the survey. Specific data about the informant's school were collected so that school and class size, information about the administration and teachers, and other pertinent factors could be considered when designing the second part of this study.

The fifth survey section offered each respondent an opportunity to comment on and delineate other factors that were perceived as contributing to the increase in CTBS composite scores. This was a particularly important section for it provided the respondent an opportunity to reflect upon the experiences of the school in relation to the increased scores.

Administration of the Survey

The 31 schools in the sample were sorted according to provincial school board. A letter was drafted to each school board superintendent asking for support in conducting this research project and assistance in distributing the survey to principals. Accompanying each letter was an unsealed, stamped envelope addressed to each of the potential respondents. This survey package contained three items: a letter to the administrator explaining the purpose of the survey and suggesting a return date; the survey; and a return envelope ready to be mailed. It was requested that the superintendent peruse the contents of the envelopes, seal them, and place them in the mail.

The superintendent's package was mailed on May 22, 1992 and the requested return date for the surveys was Friday, June 12, 1992. A copy of the survey has been included in Appendix A.

Data Analysis Procedures

Upon receipt of the completed surveys, data were analyzed. The distracters, "definitely agree" and "mostly agree", were recoded and were given a new variable name, "agree". The distracters, "definitely disagree" and "mostly disagree", were recoded and were given a new variable name, "disagree". Any statements written in a negative form were recoded so that all data could be compared in its positive form. The data were analyzed to obtain frequencies, percentages, means, and standard deviations.

Any item whose frequency of response was 85% or greater was isolated. These items were regrouped into new categories and were used to guide the design of the second part of the research.

The Second Research Question

The second research question was intended to further explore the principals' perceptions of how the factors identified by most of the respondents contributed to the increased CTBS composite scores. This research question is stated:

According to individual school principals, how do the lactors identified on the initial survey contribute to the increase in the mean grade six compusite scores on the <u>Canadian Tests of Basic Skills</u> in their respective schools during the period between 1985 and 1991?

The Case Study

According to Merriam (1988), case study research in education "should be used when certain questions are raised about a phenomenon and when a certain end product is desired." (p. 36). Yin (1984) lists five components of a case study research design that include: "a study's questions; its propositions, if any; its unit(s) of analysis; the logic linking the data to the propositions; and the criteria for interpreting the findings" (p. 29). This study followed the methodology described by Yin.

Case Study Propositions

"Each proposition in a case study directs attention to something that should be examined within the scope of the study" (Vin, 1984 p. 30). The analysis of the survey data revealed that a high percentage of respondents (greater than or equal to 85%) agreed with 13 statements from the initial survey. These statements were used as a basis for identifying five propositions for this case study:

- The principal felt that students have displayed a positive change in attitude since 1985 and that this positive attitude has contributed to the positive increase in CTBS composite scores.
- The principal left that the school is characterized by innovation and change and that these innovations and change processes have contributed to the positive increase in CTBS composite scores.

- The principal felt that the school promotes "resource-based learning" and the students are actively involved in resource based learning activities. This focus on resource based learning has contributed to the positive increase in CTBS composite scores.
- The principal felt that the public has a positive perception of life in the school and that this perception of school life has contributed to the positive increase in CTBS composite scores.
- 5. The principal felt that the parents of students in the school feel that the principal is an accessible administrator and the parents are aware of the school's programs and accomplishments. These public relation initiatives have contributed to the positive increase in CTBS composite scores.

Unit of Analysis

The unit of analysis has been called "the case" by many researchers (Yin, 1984; p. 31; Merriam, 1988; p.44). Merriam states, "the unit of analysis or "the case', can be an individual, a program, an institution, a group, an event, a concept" (p. 44). The focus of this case study was the principal's perceptions of the factors which contributed to the positive changes in the CTBS composite scores over a six-year period. The principal's perceptions, therefore, are the units of analysis in this study.

The Logic Linking the Data to the Propositions and Criteria for Interpreting the Findings

Merriam (1988) states, "data analysis is the process of making sense of one's data" (p. 127). There are "several dimensions of data analysis including analysis during data collection, the devising of categories, and the building of theory" (Merriam, p. 123).

Analysis during data collection.

Merriam (1988) advises that data analysis and data collection "should be a simultaneous process" (p. 123). Merriam continues,

The process of data collection and analysis is recursive and dynamic. But this is not to say that the analysis is finished when all the data have been collected. Quite the opposite. Analysis becomes more intensive once all the data are in, even though analysis has been an ongoing actiwity. (Werriam, 1988, p. 123)

Four separate interviews were conducted to collect data. A prepared

questionnaire was used to guide these interviews. Supplementary questions

were inter-dispersed in the conversation to clarify responses.

The devising of categories.

Vin (1984) states, "the propositions would have shaped the data collection plan and therefore would have given priorities to the relevant analytic strategies" (p. 100). Once the data from the four interviews were gathered, an outline of each interview was generated. The outline was organized topically using the five categories identified during the analysis of the survey data. This method is supported by Merriam (1988) who suggests that once the data have been gathered, it should be organized "topically or chronologically" (o. 131). Through a process of note-taking, reading, and rereading, "the notes are developed into a primitive outline or system of classifications into which data are sorted initially" (Merriam, 1988, p. 131). The outlines were scrutinized to compare the individual units of information searching for recurring regularities or themes. These common threads were related to the stated propositions of the research.

Building of theory.

The intent of this study was to explore a phenomenon. If unifying themes or recurring regularities were to emerge, then the process of theory building could begin as other researchers conduct related studies. Only then could a theory be explored.

The Interview Strategy

Vin (1984) suggests three types of case study interviews: open-ended interviews in which the , terviewer seeks to identify facts and opinions about an event; focused interviews in which the interviewer has but a short period of time to collect data and is therefore more likely to follow a certain set of questions while seeking open-ended answers; and surveys which contain structured questions and look for specific, closed responses (pp. 83-84). Given the nature of a school principal's duties during school time, the amount of time that can be reasonably committed to an interview of this type, and the purpose of this study, it was decided that a focused interview would be utilized with the author acting as interviewer.

Development of the instrument.

In focused interviews, "interviews may still remain open-ended and assume a conversational manner, but the interviewer is more likely to be following a certain set of questions derived from the case study protocol" (Vin, 1984, p. 83). Given the 45 minute duration of each interview, a list of questions based on the five categories identified in the survey was generated to use as an interview guide. The sole intent of this interview guide was to keep the interview focused while allowing the interview style to remain open-ended, within the pre-determined structura.

Sample Selection

Four school principals were interviewed in this study. Goetz and LeCompte (1984) refer to the method used to select them as criterion-based selection.

Criterion-based selection requires that the researcher establish in advance a set of criteria or list of attributes that the units for study must possess. The investigator then searches for exemplars that match the specified array of characteristics. (p. 73) Merriam (1988) suggests that this study's sample r an be described as purposive, nonprobabilistic, and criterion-based. Merriam states, "Purposive sampling is based on the assumption that one wants to discover, understand, gain insight; therefore one needs to select a sample from which one can ler m the most" (p. 48).

Given the purpose and exploratory nature of this study, a set of criteria was developed to choose the sample for the study. Four school principals were chosen based on the following criteria:

Primary criterion.

 The change in the grade six composite scores between 1985 and 1991 is a positive 15 points.

Secondary criteria.

Five secondary criteria were used to identify the sample:

- 1. The principal of the school participated in the initial survey.
- The principal has been an administrator in the school since the most recent testing year (1991).
- The sample includes principals who are administrators in schools that come from various community settings. There must be representation from both rural and urban settings.
- At least one principal is an administrator in a school that can be categorized as a "small school" by definition.
- The schools are accessible to the researcher given the time of the year and personal resources.

Administration of the Case Study

The case study was administered in four distinct phases.

Superintendent contact.

On February 6, 1993, a letter was mailed to three superintendents of education in the province of Newfoundland representing the four respondents identified for the sample. The letters requested permission for the researcher to meet with and to conduct an interview with the four principals. Permission was granted by each superintendent.

Principal contact.

The researcher contacted each principal by telephone to explain the study, to gain support, and to establish a convenient interview time. Following the telephone conversation, a letter confirming the time and the date of the interview and a copy of the questionnaire was mailed to the principal. In each circumstance, the date of mailing was two weeks prior to the interview date.

Interview.

All interviews were conducted in the principals' offices in the school identified in the case study sample. The principals were encouraged to alaborate on their responses to the questions.

Data verification.

Following the interviews, the audio tapes were transcribed in note form. A copy of these notes, and a certificate verifying the authenticity of the transcript were mailed to the respondents. Principals were encouraged to read the notes, make any corrections, and provide any additional information that would contribute to the study. The author received the annotated transcripts and certificates by return mail.

CHAPTER FOUR

Report and Analysis of Results

Introduction

The purpose of this study was to explore an existing phenomenon identified when mean composite scores on the <u>Canadian Tests of Basic Skills</u> (hereafter referred to as CTBS) were compared. By administering a survey and interviewing four elementary school principals using case study research methodology, the researcher sought to find similarities and differences in principals' perceptions of factors that could contribute to the composite score increases.

Organization of the Findings

The study was conducted in two distinct phases. In the first phase, a survey was administered. In the second phase, cuse study interviews were conducted.

In the first phase, 31 surveys were mailed to school principals whose schools were identified by the researcher as registering increases in mean grade six composite scores on the CTBS between 1985 and 1988 and again between 1988 and 1991. In the second phase of the study, four school principals representing schools selected using the selection criteria identified in chapter three, were interviewed in one 45 minute session each. During the interview, five specific topics were discussed: student attitudes toward school, innovation and planned change, resource-based learning as an innovation, public perceptions of school life, and principal support and accessibility. Each respondent was given the opportunity to discuss any aspect of these topics that they felt contributed to the increase in CTBS scores. At the end of the interview, the respondents were invited to share any other factors which could have contributed to the composite score increases. Demographic data were gathered on the survey form in the initial phase.

The interviews were conducted by the author in the school principals' offices. Each interview was recorded on audio tape with the permission of the respondents; the tape was transcribed in note form; and the transcript was returned to the respondent for verification, changes, and certification that the transcript was an accurate reflection of the principal's statements. Each interview lasted approximately 45 minutes.

For reporting convenience and anonymity, the four principals who agreed to participate in the case studies will be referred to as P1, P2, P3, and P4. The four schools will be referred to as S1, S2, S3, and S4. P1 is the principal assigned to S1; P2 is the principal assigned to S2; P3 is the principal assigned

to S3; and P4 is the principal assigned to S4.

Phase One: Survey Information

The survey was mailed to superintendents representing 31 school principals in the province of Newfoundland in May 1992. Of the surveys, 29 reached the respondents at that time. One superintendent held back two surveys until September 1992. These were completed and returned at the end of September 1992. The total number of surveys returned was 23 which translates into a 74% return rate.

Each respondent was provided the opportunity to relate personal perceptions of the factor⁻⁻ that contributed to the increased mean composite scores in their schools. In this section, one respondent indicated that the CTBS was administered using a method that differed from the standardized procedure specified ⁻ the documentation accompanying the tests. As a result, this survey was disqualified leaving the number of qualifying respondent surveys at 22.

Part I: Community Changes

This section of the survey explored factors which may have had an impact on the population of the school, but are not directly controlled by decisions made by school personnel. Table 2 is a report of the survey

statements and the percentage of respondents who agreed or disagreed with those statements.

The data presented in Table 2 indicated that further exploration of community factors would not shed light on the observed phenomenon.

Table 2

Percentage of Respondents Agreeing and Disagreeing with Statements

	Percentage of Responses (n/22)*100		
Survey Statement: Part I	Agree	Disagree	
More parents are working now than in 1985.	54.55	45.45	
Many families have left the communities we serve over the past seven years.	22.73	77.27	
Significant numbers of families have moved into the area since 1985.	18.18	77.27	
Since 1985, our students are more actively participating in community activities.	45.45	54.55	
Since 1985, community facilities (such as arenas, swimming pools, community centres, etc.) have been built or upgraded.	45.45	54.55	
A significant number of families with professional backgrounds have moved into our area.	22.73	77.27	

Part II: Curricular Changes

The curriculum is the course of study which, directed by the Department of

Education, guides content of instruction in the classroom Curing the seven

years, many new programs and curriculum documents were made available to teachers. In this part of the survey, respondents were given the opportunity to agree or disagree with statements that reflected curricular changes. Table 3 (Appendix B) is a report of those findings.

Any statement whose frequency of response was greater than or equal to 85 percent was accepted as a factor which principals felt was common to the schools demonstrating increases in composite scores. These statements were isolated, grouped by theme, and were assigned appropriate headings: <u>Student</u> <u>Attitudes Toward School</u>, <u>Innovation and Planned Change</u>, and <u>Resource-based</u> <u>Learning as an Innovation</u>. The headings served as guides for developing the interview questionnaire used in phase two of the study.

Part III: Community Factors

Student learning is influenced by forces which exist in the home, the school, and the community. When the members of the community support the activities of the school, it is perceived as legitimate. Sergiovanni (1991) states,

principals lace the problem of legitimizing what they do and how their school functions in the eyes of important publics. Let's face it - if the school bard does not believe that a school is successful, problems will arise regardless of the facts attesting its success. The same can be said for perceptions by parents and other interest groups. It is important for schools to be perceived as legitimate, and much of what the principal does is to seek this environmental legitimacy. (p. 85) An awareness of the significance of these factors and a creative, active method of harnessing the power of these influences might affect the students' drive to perform well academically. The third part of the survey provided statements about how the school shares its accomplishments with the community and how the principal views his/her role in communicating with members of the school's community. Table 4 (Appendix C) lists the statements and the frequencies of response.

Any statement whose frequency of response was greater than or equal to 85 percent was accepted as a factor which principals felt was common to the schools demonstrating increases in composite scores. These statements were isolated, grouped by theme and were assigned the appropriate headings, Public Perceptions of School Life and Public Support and Principal Accessibility. These themes served as guides for developing the interview questionnaire used in phase two of the study.

Phase Two: Case Study

Section four of the survey was used to gather demographic information about the principals and their respective schools. By combining the data gathered from this part of the survey with the data received from the Department of Education, the sample of four principals was selected for phase

two of the study. Tables 5, 6, and 7 report the pertinent data used for the selection of the respondents.

One requirement of the schools selected for phase two of the study was that the difference between the Grade VI mean composite scores between 1985 and 1991 was at least 15 composite points. Table 5 reports the mean composite scores and the difference from the first testing identified in the study and the most recent.

Table 5

	Grade VI Mean Composite Scores - CTBS			Mean Composite Score Increase from 1985 to 1991
School	1985	1988	1991	(v1991 - v1985)
S1	24	32	40	16
S2	26	36	44	18
S3	36	48	54	18
S4	12	21	40	28

Grade VI Mean Composite Scores on CTBS and Increase Organized by School

Another criteria for the selection of schools was that one school had to be, by definition, a small school. The Government of Newfoundland Department of Education sponsored a "Small Schools Project" in 1987. In the final report, recommendation 4.1 states. That a 'small school' be defined as

- (a) each primary and elementary school in which the enrollment divided by the number of grades is not greater that twelve.
- (b) each all-grade, central or regional high school in which the enrollment divided by twenty-five is not greater than the number of grades in the school. (Riggs, 1987, p. 38)

S4 is a primary and elementary school whose enrollment was 48 students in 1991. The school offered six grade levels, grades one through six. The quotient, found by dividing the enrollment by the number of grades, qualifies S4 as a small school by definition. Table 6 reports pertinent population data of the four schools selected for the study.

Table 6

Student Populations

School	Grade VI Population		School Population			
	1985	1988	1991	1985	1988	1991
S1	45	41	40	561	530	533
S2	44	42	30	280	284	247
S3	95	82	92	702	686	738
S 4	11	11	9	57	57	48

Another criteria for selecting the sample was that all principals had to have been the principal in the school during the last year that the CTBS, was administered. The most recent year of testing was 1991. In each case, the respondent was the principal during that year. Table 7 reports pertinent principal employment and qualifications data.

Table 7

Principals' Qualifications and Employment Data

Principal		Respondent Served as Principal in Testing Year			
	Degree(s) Held	1985	1988	1991	
P1	B.A. B.Ed. M.Ed. (Admin.)	No	Assistant Principal	Yes	
P2	B.A. B.Ed. M.Ed.(Teach.)	No	No	Yes	
P3	B.A. B.Ed. M.Ed. (Admin.)	Yes	Yes	Yes	
P4	B.A.(Ed.)	No	No	Yes	

The Four Schools

<u>S1</u>.

S1 is located in an urban area and provides programming for students in kindergarten to level III. There have been relatively few staff changes over the past number of years particularly in the primary and elementary areas. Most staff changes occur in the high school specialist areas. There is a full-lime resource teacher on the staff and an active Parent/Teacher Association in the school. The school plant is used as a community recreation centre after school hours. S2.

S2 is located in a coastal community. The school board recently closed an oldur school and constructed a new facility able to meet the educational needs of the children in the area. As the new school was being planned, there was a resurgence of parental support. The school board sought and received input from the members of the community as plans for the new school were being made. The students moved into the new building during the 1988-89 school year. The new school provides programming for students in kindergarten to grade eight.

The faculty of this school is quite young by Newfoundland standards. The principal reported that the average age of the teachers is 32 years. Four stall members possess Master degrees and three others are enrolled in graduate programs. Most teachers reside outside the communities served by the school.

53.

S3 is a large school located in a coastal town. The school provides programming for students in kindergarten to grade seven. The teaching faculty is older by Newloundland standards. Many teachers obtained their teaching credentials by completing distance education and "off campus" courses. There has not been much teaching staff turnover in the recent past.

Parent volunteers work in the school supporting the work of teachers and students. Recently, a Parent/Teacher Association representing two schools in the town, the primary/elementary school and the intermediate/high school, was disbanded due to leadership conflict.

<u>\$4</u>.

S4 is located on the side of a hill overlooking a small fishing community and an enclosed harbour. Behind the building, there is a wooded area that separates the school from the closest house. During the interview with the school principal, ski-doos could be heard travelling up and down paths cut through these woods. G4 serves two communities and is, by definition, a small school. The school plant has been upgraded in stages over the past number of years.

The staff, consisting of one teaching principal, two full-time teachers and one half-time teacher, teach in multi-graded classrooms. The principal takes responsibility for grades five and six, supervises the use of the resource centre, and completes administrative dulies after school hours. Under normal circumstances, the school offers the complete kindergarten to grade six program. This year, due to a lack of enrollment, the kindergarten program is not being offered. Staff meetings do not occur at regularly scheduled times throughout the year. Instead, pertinent discussions take place during the lunch or recess breaks.

The principal reported that the school is slated to close in two years. Parents of students in the school have participated in school board meetings to discuss future plans for the students.

Student Attitudes Toward School

<u>S1</u>.

Six years ago, S1 began piloting a school improvement program that was initiated by its school board in response to the effective schools literature. The staff and administration consulted and decided that the focus of the school improvement effort in the first year would be self-esteem. The principal reported that although students began to feel better about themselves, student academic achievement did not improve significantly. Recognizing that students who feel good about themselves are more likely to strive to achieve, the administration and teachers felt that a change in school improvement direction would be beneficial. In the second year, the focus broadened to include academic achievement. Other school improvement efforts have since been added. These have included whole language in the primary and elementary grades and a daily compulsory "stop everything and read" program for students in kindergarten to grade six.

<u>S2</u>.

P2 felt that the building of a new school positively affected the attitudes of the students. As members of the community worked together to plan the new facility, students' attitudes toward school changed.

P2 reported that families in the community had been negatively attected by a downturn in the fishery. Although the downturn is a relatively recent phenomenon, the community felt the impact years before the federal government publicly recognized the crisis. Traditionally, members of the community and students had regarded the school merely as a place to spend time until gainful employment could be secured in one of the community's fishery related industries. This downturn signalled a change in perspective. Parents of students realized that employment prospects for their children were dim and a good education guaranteed gainful employment outside the community. As a result, there was a positive change in parental support for and student attitudes toward education and the school.

<u>S3</u>.

P3 explained that student attitudes became more positive for a variety of reasons:

 Parents of students are themselves graduates of the school and therefore have a positive attitude toward the facility and the programs
offered within it. This attitude is passed on to the students.

- Teachers are interacting with students more than ever before and are building a better understanding of their individual needs.
- Students do not socialize outside the school environment. Televisions and video games have effectively placed socialization patterns in disarray. The only opportunities for socialization occur in school.
- For most students, the school has become an extension of the home. It is a happy place.
- Teachers are motivating students indirectly. This motivation improves self-esteem and student attitudes toward school.
- Teachers have responded to changing times. Student attitudes and achievement are linked directly to teacher attitudes and comfort levels in the school.

<u>S4</u>.

P4 and the former principal set out to build a relaxed and open atmosphere in the school. They have been successful in fostering the feeling in students that the school belongs to them and that they should take pride in it. Students are comfortable in approaching teachers about any matters of concern. The school is used as a central gathering point - a use that is encouraged by P4.

Common factors that affect basic skills.

All principals indicated that student attitudes have become more positive during the period from 1985 - 1991. In each case, the principal attributed the change in attitude to manipulation of the status quo by changes at the school level. S1 was involved in a school improvement initiative whose locus was self esteem and student achievement. S2 was involved in the planning and the building of a new school as the lishery faced crisis. S3 reported that teachers have become more aware of the needs of students and recent methodologies. S4 deliberately manipulated the atmosphere in the school to make it a conducive place for teacher/student interaction.

All principals indicated that their school is used by the community as a central gathering point. Teachers seem to be in contact with students for a greater period of time. This extension of contact time is not always curriculum related.

P1, P3, and P4 reported that teachers have responded to changing times and curriculum by upgrading their skills and adapting teaching methodologies. They have ensured that students feel comfortable in the school setting and are using newer classroom management techniques. (It should be noted that on the average, the teachers in S2 graduated from teacher training programs more recently than teachers in other schools and have entered a changing educational setting equipped with more recent attitudes and methodologies.)

Innovation and Planned Change

S1.

The school embarked on a school improvement project which has been described in the section entitled, <u>Student Attitudes toward School</u>.

Primary teachers attended a professional development workshop in "whole language". The teachers were enthusiastic about the methodology and requested that the school adopt and implement a primary whole language program. The principal supported the teachers' request. Since the implementation of this methodology, a new whole language reading program entitled, <u>Books Alivel</u> was introduced for use in grades four, five, and six.

In kindergarten to grade six, inclusive, the students and teachers embarked on yet another reading program. For 20 minutes daily, students read without interruption. Other facets of the innovation include regular home reading checks, reading promotions, and student rewards.

P1 reported that teachers were enthusiastic about the innovations implemented in the school. Teachers were involved at each stage of the implementation process.

S2.

P2 reported that the introduction of a special needs policy improved the basic skill competencies of special needs students. Students whose basic skill competencies were low required extra support from the teachers in the school. Initially, there was concern that programs for these students would be delivered in the regular classroom - a practice that differed from other remedial models. These concerns were successfully addressed and special needs students began to receive individual programs in the classroom. It should be noted that four teachers possess specialized training in developing individual program plans for special needs students.

<u>S3</u>.

S3 has been involved in many innovative program changes. P3 reported that:

- 1. teachers are planning cooperatively;
- students are preparing units of work for presentation to peers in other classes;
- school personnel have written and are continuing to write, modiy, and implement policies that address special circumstances in the school (These policies include: (a) <u>Tragic Response Policy</u>: (b) <u>Special Needs</u> <u>Policy</u>: (c) <u>Assertive Discipline Policy</u>: and (d) <u>Teacher Advisory Policy</u>); and
- the school has piloted various models of parent orientation seminars in an effort to identify the approach that best communicates information

about the school, its programs, its teachers, and its facilities.

<u>S4</u>.

P4 reported the following three innovations in his school:

1. an on-going, deliberate attempt to instill a family atmosphere,

- a new reading and writing program that addresses basic skill competencies and habits (Teachers now use a variety of creative activities that encourage students to read and write more.); and
- provision for extra classroom time to address low student scores in solving non-routine problems.

P4 noted that the teachers are constantly involved in innovative practices because of the nature of their school. The curriculum prescribed by the Department of Education lends itself best to single graded classroom settings. The multi-graded classroom requires a .nore flexible, creative, and innovative approach to program delivery.

Common factors that affect basic skills.

P1, P2, P3, and P4 reported that they have been involved in some form of innovative practice. Teachers have been involved in the recognition and the communication of a need, the development of an innovation that responds to that need, and the activity that occurs after the implementation of the innovation. The four principals reported that the innovative programs, policies, and practices are evolving and can be plotted at different points along a planned change continuum.

These principals reported that they have implemented innovations that address a need to improve the basic skills of their students.

Resource-based Learning as an Innovation

<u>S1</u>.

While P1 served as vice-principal in the school, the principal attemptedto implement resource-based learning in all sections of the school. Since P1's appointment as principal, the use of resource-based learning has been inconsistent in various grade levels. The administration is actively attempting to raise teacher awareness and commitment to the methodology. In light of this, P1 reported that a special effort has been made to monitor the implementation of resource-based learning, keep in contact with the resource teacher, and support the growth of the methodology in the school.

Computers are housed in two separate rooms outside the school's resource centre. The principal reported that although primary and elementary teachers have used computers to provide resource-based instruction, high school students use the computer room facilities and equipment to complete prescribed computer courses. Barriers for continued primary and elementary use of the computers include the initexible computer room schedule and the lack of a computer literate teacher who is enthusiastic about the instructional use of computers at the primary and elementary levels. The principal has recognized these barriers.

S2.

Before the Department of Education released their 1991 policy publication entitled, <u>Learning to Learn</u>, the principal and the resource teacher attended a three-day resource-based learning symposium. Upon their return to the school, they planned and delivered to their colleagues two resource-based learning workshops. As a result of these workshops:

- teachers work in conjunction with the resource teacher to plan and deliver resource-based units;
- students use computers to write stories, research information, and practice skills (P2 stated that students are motivated when they use computers and good educational software provides immediate reinforcement of basic skills.); and
- students in grades seven and eight receive specific instruction in wordprocessing and the use of spreadsheets.

P2 reported that the basic skills of students are positively affected when they are completing resource-based learning units. As more time is used to complete these units, students become more proficient in the basic skills tested in the visual materials, mathematics, and language sub-tests of the CTBS

<u>S3</u>.

In the adoption, planning, and implementation stages of resource-based learning, school administrators worked very closely with the program coordinator responsible for resource-based learning. Many teachers adopted the methodology and became role models for others to get involved. When the policy document entitled, Learning to Learn was released by the Department of Education in 1991, the principal and the program coordinator already had experienced success with implementing resource-based learning and were asked to become part of a large, district committee struck to devise an actionplan for district-wide implementation. P3 reported that the committee met once and was disbanded because school board personnel lacked the ownership required for implementation in all schools throughout the school district. In light of this failed attempt at district-wide change, the school adminstration, with the support of the program coordinator and enthusiastic teachers, continued to implement its own action plan. P3 reported that most teachers in his school have "bought-in" and are presently using the resource-based learning methodology.

<u>S4</u>.

P4 reported that in order to deliver a successful program to students in multi-grade classrooms, teachers have to use some form of resource-based learning. Because the school lacks the substantial number and variety of resources required to implement a good program, funding from a small school budget has been allocated yearly to ourchase additional resources.

Students are allowed to use two computers in the school - an Apple computer situated in the primary classroom and an I.B.M.-compatible computer with a CD-ROM drive on loan from the school board office. A staff member has located and gained possession of computer parts and is trying to assemble the pieces to provide another computer for student use in the school.

P4 reported that "resource-based teaching" is going on in the school. The principal deliberately referred to the practice as resource-based teaching not resource-based learning. It was argued that because students are not directly involved in choosing the direction of their own learning, true resourcebased learning has not been implemented. Resource-based teaching better describes what happens when teachers have a finite number of resources that can be used to deliver the prescribed curriculum.

Common factors that affect basic skills.

P1, P2, P3, and P4 reported that their schools were involved in

104

implementing resource-based learning before the Department of Education published its policy document entitled, <u>Learning to Learn</u> in 1991. P1, P2, and P3 reported that a school-based resource-based learning action plan was put into place. P4 reported that the nature of the school and available resources required teachers to use a modified version of resource-based learning.

The four principals have a personal commitment to resource-based learning and are actively involved in the implementation process.

P2, P3, and P4 are committed to purchasing computers and software for student and teacher use. P1 e-pressed an awareness that the school must refocus its school improvement efforts to include resource-based learning.

The four principals support and encourage the use of resource facilities during class time for varied resource-based learning activities.

Public Perceptions of School Life

<u>S1</u>.

The discussion of this aspect of the study was limited in scope.

However, it was expressed that the administration of the school plays an active role in shaping the public perceptions of school life through communication and public relations initiatives.

At various times during the interview, the following two points became clear:

- Parents feel that the school is serious about the education of their children and is striving to do its best for them.
- Parents perceive that their opinions about the school count. They have confidence in relating their ideas to the school principal.

S2.

The principal reported that the public has a positive perception of the school. This view is communicated directly to the principal by members of the community, as second-hand information through relatives, and from members of the Parent/Teacher Association. The public praises the work of the teachers and recognizes their dedication to the education of their children.

The principal pointed out that some members of the public associate the number of cars in the parking lot early in the morning and late in the afternoon with staff dedication. Teachers do work long hours, do arrive early at the school, and do remain at the school for hours after students are dismissed. This positively affects public perceptions.

S3.

The principal reported that generally, parents now have a positive perception of the school. This is the result of an active campaign to change the habits and attitudes of the public. In particular, a story was related about school concerts. Traditionally, concerts in this community had always started late. Audiences arrived after scheduled curtain times and still were able to see complete performances. Children had traditionally gone home tired and had trouble getting out of bed to go to school the next day. Early in the principal's tenure, he resolved this problem. As always, he scheduled a concert taking into account fhe age of the student performers, the length of the concert, and what he thought was appropriate bed-times for the children. Although the curtain time was set and published, he started a concert early. Late-comers missed parts of their child's performance. As a result, audiences now arrive at all school functions early to avoid disappointment.

The principal reported that although legitimate complaints are welcomed, the number received has been minimal. When a complaint is received, the principal acts as a judge. He listens to the complaint, considers the circumstances, and places blame where it belongs. He believes that this straight-forward approach has aided in fostering a positive public perception of the school. In the past, complaints were aired and discussed in the lobby of the post office.

<u>\$4</u>.

P4 reported that he resides in one of the small communities served by the school, is well-known by most, is on a "first-name basis" with most, and

107

socializes with some parents. Parents do not hesitate to telephone the principal at home to question school procedures, decisions, or program-related issues. Generally, members of the community have a positive perception of the school.

Common factors that affect basic skills.

In all four cases, the principals related information and incidents illustrating that members of the public support the school, its staff, and its programs.

These principals reported that parental support has had a positive effect on student basic skill competencies.

Public Support and Principal Accessibility

<u>S1</u>.

P1 reported that parents are able to support their children more than they have ever been able to before. This new generation of parents possesses more education and is more able and willing to supplemant the work of teachers by supervising homework assignments. This has had a substantial effect on basic skill competencies of students.

Many factors were discussed that illustrate positive public support for the school, its programs, and its policies:

1. The number of telephone calls that register complaints has decreased in

number. Parents may not always agree with decisions that the school is forced to make, however, they do support the school's programs.

- There has been tremendous support for fund-raising initiatives in the school.
- Parent attendance at school functions has been extremely high. The principal reported that teachers who have been on the staff for many years attest to a change in school function support.
- Parents make telephone calls to the school looking for ways in which they can support the education of their children.
- Radio stations have been very supportive and have publicized school events.

The principal stated that the public supports him in his job. He hesitated to comment on how he was aware of the support. P1 reported that he makes a special effort to accommodate meetings with members of the public when requested. He feels that this use of administration time is valuable.

<u>S2</u>.

P2 reported that members of the public support him. This perception is based on second-hand information that has been shared with relatives who reside in the community. The principal is seen as a "regular person" rather than an authoritative ruler. Members of the community feel at ease when approaching him with a problem. It was reported that former principals did not always enjoy this kind of community support.

Parental support for homework has increased. There has been more of an effort on the part of parents to ensure that homework assignments are completed correctly. There is still room for further support in this area.

Parents supported the school board when designing and planning the new school building.

53.

Parents can request meetings with the principal whenever a need arises. The principal grants these meetings when they are requested.

There is an active parent volunteer program in place in the school.

<u>S4</u>.

P4 reported that there is support for him among the parent population. This support was reported in a variety of ways:

 At a recent meeting to discuss the eventual closure of the school, the parents in attendance raised a concern about the fate of the principal once the school was closed. (While relating the incident, the principal had to be prodded to provide the details. It seemed that parents support and respect the professionalism of the principal and the staff and are generally concerned for their future well-being.)

- There has been an attempt to increase the amount of time parents spend in the school. Periodically, parents come into the school to serve hotdogs to students at lunchtime. Other parents shelve books in the resource centre.
- The principal has developed good rapport with the public. Members of the community feel free to approach the principal about educational matters at any time - ten miles in the country, in the school, or at community functions.

It was reported that a small number of parents do not support the school. The principal recognizes this and has actively campaigned to regain their support.

Parents' efforts to support the education of their children receive positive reinforcement from the school. This reinforcement translates into greater educational support for the student and higher basic skill competencies.

Common factors that affect basic skills.

The four principals reported a perception that the public was supportive of them in their administrative roles.

P1, P2, P3, and P4 reported that parents are more able to support students in their educational endeavours. P1, P3, and P4 reported that parents of younger students support the school more than parents of older students. P1 and P3 reported that mar:: young parents are themselves graduates of the school. P1, P2, and P3 reported that increased parental support translates into increased basic skill competencies.

The four principals encourage parents to participate in school activities and frequently communicate with them by means of memos, newsletters, or a combination of both.

Other Factors

<u>S1</u>.

The principal reported that the following points must be considered when looking at factors that contribute to basic skill increases:

- Teacher attitudes toward the community and teacher perceptions of student capabilities have changed positively over the past few years.
- Teachers have a positive impression of their own professional accomplishments in the school.
- 3. The school is a community school and is supported by all.
- Although positive progress has been made in all aspects of the school's program, all staff members recognize that the school must continue on in its quest for improvement.

S2.

The principal reported that the following must be considered when looking at factors that contribute to basic skill increases:

- There used to be too much opportunity for employment in the community. There is a new realization that education is a vehicle that will ensure future employment.
- 2. The teaching staff in the school is highly qualified and competent.
- Teachers strive for personal excellence. This excellence is reflected in the academic achievement of students.
- The principal supports the professional decisions of members on the staff.
- The principal sees himself as an instructional leader who uses democratic principles when making decisions.
- 6. The principal and vice-principal work well together.

53.

The principal reported that the following must be considered when looking at factors that contribute to basic skill increases:

 The principal takes a "hardline" approach to school administration and demands the best from the teaching stalf while creating a balance of collegial rapport and professionalism.

- 2. The principal and the vice-principal work closely as a team.
- The principal professes a clear vision for Canadian education and seeks to ensure that students leaving the school are given the tools required to be a part of that vision.
- The principal identifies many objectives and goals for the school yearly and actively sets out to ensure that they are met.

<u>\$4</u>.

The principal reported that the following must be considered when looking at factors that contribute to basic skill increases:

- 1. The principal is an instructional leader on the staff of the school.
- There is a community atmosphere felt by all who have contact with the school.

Common factors that affect basic skills.

The four principals felt that teachers have a commitment for their students and for their own professional development. This personal commitment has had a positive effect on the basic skill competencies of the students.

CHAPTER FIVE

Conclusions and Recommendations

Summary

This study was conducted to explore perceptions of elementary school principals as they relate to factors that have contributed to local increases in grade six composite scores on the <u>Canadian Tests of Basic Skills</u> (hereafter referred to as CTBS) over a period of six years. The study was conducted in two distinct phases.

Included in the first phase was the distribution and collection of a survey and the collation and analysis of the data. The survey contained statements of a variety of factors that could have contributed to the increased performance of students on the CTBS. The principal of each school that registered a positive grade six composite score increase between 1985 and 1988 and again between 1988 and 1991 was sent a copy of the survey for his/her consideration. Twenty-three of thirty-one surveys were returned for analysis.

The data collected from the surveys indicated that, generally, principals did not attribute the positive increases to demographic changes in the communities they served. Instead, the data reflected general principal agreement of factors relating to five specific themes: <u>Student Attitudes toward</u> School. Innovation and Planned Change. Resource-based Learning as an Innovation. Public Perceptions of School Life, and Public Support and Principal Accessibility.

The second phase of the research included four case study interviews. Four principals who had participated in the survey sample agreed to participate in an interview designed to explore the five common themes identified from the survey data. The qualitative data collected during the interviews were analyzed to determine common factors that principals felt had contributed to the increased performance of grade six students on the CTBS in their schools. The interviews served to answer the primary question identified for the research:

According to individual school principals, how do the factors identified on the initial survey contribute to the increase in the mean grade six composite scores on the <u>Canadian Tests of Basic Skills</u> in their respective schools during the period between 1985 and 1991?

Conclusions

Principals generally found it difficult to articulate how any factors had contributed to the increased performance on the CTBS. However, they were sure that they and their teachers have been and continue to be engaged in practices that lead to positive changes in student achievement and more specifically to increased basic skill competency.

During the interviews, principals identified specific influences that they

felt had contributed to increased student basic skill performance and described specific interventions that they felt had contributed to the composite score increases on CTBS Not surprisingly, the four principals reported some common influences and interventions. These common factors are summarized under their appropriate theme:

Student Attitudes Toward School

- The four principals reported that student attitudes had become more positive over the course of the three testing periods.
- The four principals reported that teachers are in contact with students for longer periods of time now than in the past. This contact is not necessarily for instruction.
- The four principals reported that teachers have upgraded their professional qualifications and now use methodologies and classroom management techniques that contribute to positive attitudes toward school.

Innovation and Planned Change

- The four principals reported that the school has implemented some innovative practices and that teachers were involved in the identification, adoption, and implementation process.
- The four principals reported that they have implemented some innovation that has addressed the need to improve basic skill performance of

students.

Resource-based Learning as an Innovation

- 6. The four principals reported that their schools have implemented a form of resource-based learning. (Some resource-based learning modifications have been made to suit special school circumstances.)
- The four principals reported a personal commitment to resource-based learning and are actively involved in the implementation process.

Public Perceptions of School Life

- The four principals reported that members of the public support all aspects of the school, its staff, and its program.
- The four principals reported that positive parental support has had a beneficial effect on the basic skill competencies of students.

Parental Support and Principal Accessibility

- The four principals reported that they perceived support for their role as principal from most parents.
- The four principals reported that, in general, today's parents have a stronger academic background that enables them to support their children's educational endeavours.
- The four principals reported that they encourage parents to participate in school activities.
- 13. The four principals reported that they communicate with parents

frequently by means of newsletter, memorandum, or a combination of both.

Other Factors

14. The four principals reported that teachers demonstrate a commitment for their students and for their own professional development. This commitment positively affects the basic skill competencies of students in schools.

Summary

Principals of schools whose grade six students demonstrated an increase in basic skill composite scores on the CTBS between 1985 and 1988 and again between 1988 and 1991 agree that the following factors have contributed to the local increase in basic skill composite scores:

- The implementation of interventions aimed at fostering positive student attitudes toward school;
- The implementation of innovations that reflect current educational thought and address needs identified by teachers and administrators, working together;
- A demonstration of support by educators, parents, and members of the community for all school-wide innovations;
- 4. A positive public perception that school is beneficial for children and

active parental support for students in all educational endeavours; and

 A staff committed to the educational welfare of students and to personal professional development.

Recommendations

Based on the summary and conclusions and recognizing the exploratory nature of this study, the author makes the following recommendations:

- that subsequent research be carried out to compare schools that have registered a constant increase in CTBS composite scores with schools that have not registered a constant increase;
- that subsequent research be carried out to replicate the findings of this study following the 1994 CTBS testing period;
- that subsequent research be carried out to identify common factors that superintendents, program coordinators, teachers, parents, and students perceive have contributed to the increases in basic skill composite scores on the CTBS; and
- that subsequent researcl, be carried out to explore the same phenomenon using grade four and grade eight CTBS composite scores as a beginning point.

BIBLIOGRAPHY

- Bagley, C., & Hunter, B. (1992). Restructuring, constructivism, and technology: Forging a new relationship. <u>Educational Technology</u>, <u>32</u>(7), 22-27.
- Beare, H., Caldwell, B. J., & Millikan, R. H. (1989). <u>Creating an excellent</u> <u>school: Some new management techniques</u>. New York: Routledge.
- Bickel, W. E. (1983). Effective schools: Knowledge, dissemination, inquiry. <u>Educational Researcher</u>, 12(4), 3-5.
- Blase, J., & Kirby, P. C. (1992). Bringing out the best in teachers: What effectiv. principals do. Newbury Park, CA: Corwin Press.
- Block, A. W. (1983). <u>Effective schools: A summary of research</u>. Arlington, VA: Educational Research Service.
- Caldwell, J. H., Huitt, W. G., & Graebar, A. O. (1982). Time spent in learning: Implications from research. <u>The Elementary School Journal</u>. 82, 471-480.
- Clark, D. L., Lotto, L. S., & Astuto, T. A. (1984). Effective schools and school improvement: A comparative analysis of two lines of inquiry. <u>Educational</u> <u>Administration Quarterly</u>, 22(3), 41-64.
- Coleman, J. S., Campbell, E., Hobson, C., McPartland, J., Mood, A., Weinfield, F., & York, R. <u>Equality of educational opportunity</u>. Washington, DC: U.S. Government Printing Office.
- Cox, P. L. (1983). Complementary roles in successful change. <u>Educational</u> <u>Leadership</u>, <u>41</u>(3), 10-13.
- Crandall, D. P. (1983). The teacher's role in school improvement. Educational Leadership, 41(3), 6-9.
- Davis, G. A., & Thomas, M. A. (1989). <u>Effective schools and effective teachers</u>. Boston: Allyn and Bacon.

- Edmonds, R. R. (1981). Making public schools effective. <u>Social Policy</u>, <u>12</u>(2), 56-60.
- Edmonds, R. R. (1982). Programs for school improvement: An overview. Educational Leadership, 40(3), 4-11.
- Fletcher, K. (1986). Making schools more effective. <u>Education Canada</u>, <u>26</u>, 44-46.
- Fullan, M. (1982). The meaning of educational change. Toronto: OISE.
- Fullan, M. (1984). The principal as an agent of knowledge utilization (ku) for school improvement. In D. Hopkins & M. Wideen (Eds.), <u>Alternative</u> <u>perspectives in school improvement</u> (pp. 89-103). London: Falmer.
- Fullan, M., Bennett, B., & Roheiser-Bennett, C. (1990). Linking classroom and school improvement. <u>Educational Leadership</u>, 47(8), 13-19.
- Fullan, M., & Miles, M. B. (1992). Getting reform right: What works and what doesn't. <u>Phi Delta Kappan</u>, <u>73</u>(10), 745-756.
- Gage, N. (1984). What do we know about teaching effectiveness? <u>Phi Delta</u> <u>Kappan, 66</u>, 87-93.
- Gallivan, J. (1986). Canadian tests of basic skills. <u>Canadian Journal of School Psychology</u>, 2(2), 69-73.
- Garbarino, J., & Asp, C. E. (1981). <u>Successful schools and competent</u> students. Lexington: Lexington Books.
- Goetz, J. P., & LeCompte, M. D. (1984). <u>Ethnography and qualitative design in</u> educational research. New York: Academic.
- Government of Newfoundland and Labrador (1984). The aims of public education. St. John's, NF: Department of Education.
- Government of Newfoundland and Labrador (1988). Testing standards: Grade <u>6</u>. St. John's, NF: Department of Education.
- Government of Newfoundland and Labrador (1989). Early childhood program guide. St. John's, NF: Department of Education.

- Government of Newfoundland and Labrador (1990). <u>Challenge for</u> excellence. St. John's, NF: Department of Education.
- Government of Newfoundland and Labrador (1991). <u>Learning to learn: Policies</u> and guidelines for the implementation of resource-based learning in <u>Newfoundland and Labrador schools</u>. St. John's, NF: Department of Education.
- Government of Newfoundland and Labrador (1991). <u>Testing standards: Grade</u> <u>6</u>. St. John's, NF: Department of Education.
- Guskey, T. R. (1985). Staff development and teacher change. <u>Educational</u> <u>Leadership</u>, 42(7), 57-60.
- Guskey, T. R. (1989). Integrating innovations. <u>Educational Leadership</u>. <u>47</u>(4), 11-17.
- Guthrie, J. W., & Reed, R. J. (1991). Educational administration and policy: Effective leadership for American education. Boston: Allyn and Bacon.
- Hakim, C. (1987). <u>Research design: Strategies and choices in the design</u> of social research. London: Allen & Unwin.
- Hallinger, P., Murphy, J., & Hausman, C. (1992). Restructuring schools: Principal's perceptions of fundamental educational reform. <u>Educational</u> <u>Administration Quarterly</u>, 28(3), 330-349.
- Hall, G. E. (1988). The principal as the leader of the change facilitating team. Journal of Research and Development in Education, 22(1), 49-59.
- Hall, G.E., & Loucks, S.F. (1978). Teacher concerns as a basis for facilitating staff development. <u>Teachers College Record</u>, <u>80</u>(1), 36-53.
- Hall, G. E., Rutherford, W. L., Hord, S. M., & Huling, L. L. (1984). Effects of three principal styles on school improvements. <u>Educational Leadership.</u> <u>41</u>(5), 22-29.
- Havelock, R. G. (1971). <u>Planning for innovation through dissemination and utilization of Knowledge</u>. Ann Arbor, MI: Centre for Research on Utilization of Scientific Knowledge.

- Havelock, R. G. (1973). <u>The change agent's guide to innovation in education</u>. Englewood Cliffs: Educational Technology Publications.
- Holmes, M. (1969). School effectiveness: From research to implementation to improvement. In M. Holmes, K. A. Leithwood, & D. F. Musella (Eds.), Educational policy for effective schools (pp. 3 - 30). Toronto: Oise.
- Holmes, M., Leithwood, K. A., & Musella, D. F. (Eds.). (1989). Educational policy for effective schools. Toronto: Oise.
- Huberman, A. M. (1983). School improvement strategies that work: Some scenarios. <u>Educational Leadership</u>, <u>41</u>(3), 23-27.
- Huberman, A. M., & Miles, M. B. (1984). <u>Innovation up close</u>. New York: Plenum.
- Johnson, N. A., & Holdaway, E. A. (1991). Perceptions of effectiveness and the satisfaction of principals in elementary schools. <u>Journal of Educational Administration</u>, 29(1), 51-70.
- Joyce, B. R. (1991). The doors to school improvement. <u>Educational</u> <u>Leadership</u>, <u>48</u>(8), 59-62.
- Joyce, B., Murphy, C., Showers, B., & Murphy, J. (1989). School renewal as cultural change. <u>Educational Leadership</u>, 47(3), 70-77.
- Leithwood, K. A. (1986). Planned Educational Change. Toronto: OISE.
- Lezotte, W. L., & Bancroft, B. A. (1985). Growing use of effective schools model for school improvement. <u>Educational Leadership</u>, 42(6), 23-27.
- Loucks, S. F. (1983). At last: Some good news from a study of school improvement. <u>Educational Leadership</u>, <u>41</u>(3), 4-5.
- Loucks, S. F., & Zacchei, D. A. (1983). Applying our findings to today's innovations. <u>Educational Leadership</u>, <u>41</u>(3), 28-31.
- Loucks-Horsley, S., & Hergert, L. F. (1985). <u>An action guide to school</u> <u>improvement</u>. Alexandria: ASCD.

- MacKenzie, D. E. (1983). Research for school improvement: An appraisal of some recent trends. <u>Education Researcher</u>, <u>12</u>(4), 3-8.
- Maxwell, T. W., & Thomas, A. R. (1991). School climate and school culture. Journal of Educational Administration, 29, 72-80.
- Merriam, S. B. (1984). <u>Case study research in education: A qualitative</u> approach. San Francisco: Jossey-Bass.
- Michaels, K. (1988). Caution: Second-wave reform taking place. <u>Educational Leadership</u>, <u>45</u>(5), 3
- Miles, M. B. (1983). Uraveling the mystery of institutionalization. <u>Educational</u> <u>Leadership</u>. <u>41</u>(3), 14-19.
- Moorthy, D. (1992). The Canadian principal of the 90's. Education Canada. 32(2), 8-11.
- Nicholls, A. (1983). Managing educational innovations. London: George Allen & Unwin.
- Purkey, S. C., & Smith, M. S. (1983). Effective schools: A review. <u>Elementary</u> <u>School Journal</u>, 83, 427-452.
- Purkey, S. C., & Smith, M. S. (1985). School reform: The district policy implications of the effective schools literature. <u>Elementary School</u> Journal, <u>85</u>(3), 353-390.
- Reid, K., Hopkins, D., & Holly, P. (1987). <u>Towards the effective school: The</u> problems and some solutions. Oxford: Basil Blackwell.
- Riffle, A. J. (1987). A developmental perspective on school improvement. <u>The Canadian Administrator</u>, <u>26</u>(4), 1-5.
- Riggs, F. (1987). <u>Report of the small schools study project</u>. St. John's: Jesperson.
- Robinson, G. (1985). <u>Effective schools research: A guide to school</u> improvement. Arlington, VA: Educational Research Service.
- Rogers, E. M. (1962). Diffusion of innovations. New York: The Free Press.

- Rowan, B., Bossert, S. T., & Dwyer, D. C. (1983). Research on effective schools: A cautionary note. <u>Educational Researcher</u>, <u>12</u>(4), 24-31.
- Ruddick, J. (1961). <u>Innovation and Change</u>. Milton Keyes: Open University.
- Runkel, P. (1984). Maintaining diversity in schools. In D. Hopkins & M. Wideen (Eds.), <u>Alternative perspectives on school improvement</u> (pp. 167-187). New York: Falmer.
- Schaller, L. E. (1972). <u>The change agent: The strategy of innovative leadership</u>. Nashville: Abingdon.
- Sergiovanni, T. J. (1991). <u>The principalship: A reflective practice perspective</u> (2nd ed.). Boston: Allyn and Bacon.
- Serotnik, K. A. (1985). School effectiveness: A bandwagon in search of a tune. <u>Educational Administration Quarterly</u>, 21(2), 135-140.
- Sizer, T. R. (1985). Common sense. <u>Educational Leadership</u>, 42(6), 21-22.
- Sparks, D. (1993). 13 tips for managing change in schools. <u>The Education</u> <u>Digest</u>, <u>58</u>(6), 13-15.
- Van Maanen, J. (Ed.). (1983). <u>Qualitative methodology</u>. Beverly Hills, CA: Sage.
- Walter, J. E. (1981). Schools that work. Social Policy, 12(2), 26-29.
- Yin, R. K. (1984). <u>Case study research: Design and methods</u>. Beverly Hills, CA: Sage.

APPENDIX A - Survey

School Name:	
School District:	
Present Principal:	
School Telephone:	
Convenient Phone Contact Time:	

PART I: COMMUNITY CHANGES

Place an X in the column which best reflects your views about variations of conditions in the communities served by your school.

	Definitely Agree	Mostly Agree	Mostly Disagree	Definitely Disagree
More parents are working now than in 1985.				
Many families have left the communities we serve over the past seven years.				
Significant numbers of families have moved into the area since 1985.				
Since 1985, our students are more actively participating in community activities.				
Since 1985, community facilities (such as arenas, swimming pools, community centres, etc.) have been built or upgraded.				
A significant number of families with professional backgrounds have moved into our area.				

PART II: CURRICULAR CHANGES

Place an X in the column which best reflects your views about the changes which have occurred in your school between 1985 and 1991.

	Definitely Agree	Mostly Agree	Mostly Disagree	Definitely Disagree
Department of Education curriculum changes over the last seven years have contributed to our increase in basic skill scores.				
There have been significant policy changes implemented by our school board during the past seven years.				
There has been a new program added (such as French Immersion) which has contributed to an increase in our basic skill scores.				
We have established a successful parent volunteer program.				
Our school is characterized by innovation and deliberate, planned change,				
A resource centre has been added to our school.				
We have purchased computers over the last seven years.				
Our school has modern, up-to-date facilities and equipment.				
There have been considerable staff changes over the past seven years.				
A resource teacher has been allocated to our staff since 1985.				
Our staff members continue to upgrade their credentials by reading current journal articles and/or by taking university courses.				

	Definitely Agree	Mostly Agree	Mostly Disagree	Definitely Disagree
The average teacher certification of our stalf has increased during the last seven years.				
Our staff has become more attuned to the effects of stress and over the last seven years has become better stress managers.				
The amount of daily teacher planning time has increased over the past seven years.				
The amount of work that teachers take home with them has decreased over the past seven years.				
Resource-based learning is "alive" in our school.				
The chalk hoard is being used as much as it ever was.				
Our students spend more time working with resources than ever before.				
Our students have been engaged in an increasing number of co-operative learning activities.				
Our students are encouraged to use computers during instructional time.				
Out students take an active role in planning, their educational experiences.				
Students have a positive attitude toward the school's facilities, programs, and staff,				
During the last seven years the number of disruptive students has decreased.				

	Definitely Agree	Mostly Agree	Mostly Disagree	Definite ly Disagree
Students have become better able to stay on-task when working through activities.				
Our students today appear to be more confident and have higher self-esteem.				
Our school funding is adequate and allows us to keep up with changing educational demands.				
PART III: COMMUNITY FACTORS

Place an X in the column that best reflects your views about any outside influences which may have affected basic skills scores from 1985 to 1991.

	Definitely Agree	Mostly Agree	Mostly Disagree	Definitely Disagree
Public media messages have positively influenced attitudes toward school.				
We have a very active Parent/ Teacher Association,				
Generally, parents are playing a more active part in supervising homework activities.				
The general public is viewing our school as more successful.				
The members of our community are becoming more aware of and more supportive of the educational activities in our school.				
Parents are feeling more welcome to visit our school at any time - even when classes are in session.				
We are promoting our accomplishments publicly more than we ever did.				
The staff and administration publishes a newsletter for parents and distributes it regularly throughout the year.				
Parents generally feel that I am an accessible administrator and feel comfortable when approaching me with concerns.				
Parents actively volunteer to help prepare and/or participate in classroom activities.				
Many of our school's families have purchased computers since 1985.				

PART IV

To complete the following questions, it may be necessary to consult some past school records. Please be as accurate as possible.

In this section, we are interested in data from three separate and distinct school years. Please record the answer in the appropriate column.

	1985-86	1988-89	1991-92
Total student population.			
Grade Six student enrollment.			
Number of Grade Six classes.			
Number of elementary teachers. (Count only teachers responsible for a homeroom clasa.)			
Number of teachers on staff.			
I was the administrator of this school during these years. (Please check.)			
Number of teachers who have been in the same Grade Six teaching position over the past seven years.			

School Years

PART V

In this section, please delineate any aspects of your school which you think might have affected the increase in basic skill scores among your Grade Six students.

Thank-you for your time and assistance.

APPENDIX B - Table 3

	Percentage of Responses (n/22)*100		
Survey Statement: Part II	Agree	Disagree	
Department of Education curriculum changes over the last seven years have contributed to the increase in basic skill scores.	54.55	45.45	
There have been significant policy changes implemented by our school board during the past seven years.	72.73	27.27	
There has been a new program added (such as French Immersion) which has contributed to an increase in our basic skill scores.	31.82	68.18	
We have established a successful parent volunteer program.	36.36	63.64	
 Our school is characterized by innovation and planned change. 	90.91	9.09	
A resource centre has been added to our school.	45.45	54.55	
*We have purchased computers over the last seven years.	90.91	9.09	
Our school has modern, up-to-date facilities and equipment.	77.27	22.73	
There have been considerable staff changes over the past seven years.	31.82	68.18	
A resource teacher has been allocated to our staff since 1985.	31.82	68.18	
Our staff members continue to upgrade their credentials by reading current journal articles and/or by taking university courses.	77.27	18.18	
The average teacher certification of our staff has increased during the last seven years.	68.18	31.82	

Table 3 Percentage of Respondents Agreeing and Disagreeing with Statements

Our staff has become more attuned to the effects of stress and over the last seven years has become better stress managers.	40.91	59.09
The amount of daily teacher planning time has increased over the past seven years.	36.36	63.64
"Resource-based learning is "alive" in our school.	86.36	13.64
The chalk board is being used as much as it ever was.	31.82	68.18
'Our students spend more time working with resources than ever before.	95.45	4.55
*Our students have been engaged in an increasing number of co-operative learning activities.	86.36	13.64
Our students are encouraged to use computers during instructional time.	81.82	18.18
Our students take an active role in planning their educational experiences.	63.64	36.36
'Students have a positive attitude toward the school's facilities, programs, and staff.	100.00	0.00
During the last seven years the number of disruptive students has decreased.	68.18	31.82
'Students have become better able to stay on-task when working through activities.	86.36	13.64
Our students today appear to be more confident and have higher self-esteem.	77.27	22.73
Our school funding is adequate and allows us to keep up with changing educational demands.	22.73	77.27

Note. ' indicates a statement whose frequency is greater than or equal to 85 percent.

_

APPENDIX C - Table 4

Table 4

Percentage of Respondents Agreeing and Disagreeing with Statements

_	Percentage of Responses (n/22)*100		
Survey Statement: Part I	Agree	Disagree	
Public media messages have positively influenced attitudes toward school.	63.64	36.36	
We have a very active Parent/Teacher Association.	36.36	63.64	
Generally, parents are playing a more active role in supervising homework activities.	68.18	31.82	
*The public is viewing our school as more successful.	90.91	9.09	
*The members of our community are becoming more aware of and more supportive of the educational activities in our school.	95.45	4.55	
*Parents are feeling more welcome to visit our school at any time - even when classes are in session.	95.45	4.55	
*We are promoting our accomplishments publicly more than we ever did.	86.36	13.64	
The staff and administration publishes a newsletter for parents and distributes it regularly throughout the year.	72.73	27.27	
*Parents generally feel that I am an accessible administrator and feel comfortable when approaching me with concerns.	95.45	4.55	
Parents actively volunteer to help prepare and/or participate in classroom activities.	59.09	40.91	
Many of our school's families have purchased computers since 1985.	45.45	54.55	

Note. * indicates a statement whose frequency is greater than or equal to 35 percent. APPENDIX D - Interview Questions

INTERVIEW QUESTIONS

Student Attitudes Toward School

You have indicated that generally the students in your school have displayed a positive change in attitude since 1985.

To what do you attribute this change?

How has this change affected your students' general basic skills achievement?

In your opinion, how much of the increase in basic skills would you attribute to student attitudes toward school?

Innovation and Planned Change

You have indicated that your school is characterized by innovation and planned change.

What innovations and changes have been encouraged by you in your school since 1985?

Who has been involved in this (these) planned change(s)?

What has (have) their role(s) been?

What has your role been in this planned change?

How have your students been affected by the implementation of this (these) innovation(s)?

How has planned change affected your students' basic skills?

In your opinion, how much of the increase in basic skills do you attribute to the planned change initiatives?

Resource Based Learning as an Innovation

You have indicated that resource based learning is "alive" in your school and that your students are actively involved in a number of resource based learning activities.

What makes you believe that resource based learning is "alive" in your school?

Does or did your school have a resource based learning implementation plan? (If so, please describe the plan.)

What is and has your role been in the implementation of the resource based learning model?

How are resource facilities being used by your students?

How has this affected the basic skills of your students?

In your opinion, how much of the increase in basic skills can be attributed to resource based learning?

Public Perceptions of School Life

You have indicated that the general public has a positive perception of life in your school.

What do you think has contributed to this perception?

Have you had a direct role in shaping that perception? What leads you to that conclusion?

How have these perceptions affected the basic skills of your students?

In your opinion, how much of the increase in basic skills can be attributed to the public's perception of life in your school?

Public Support and Principal Accessibility

You have indicated that you believe that the parents of students in your school

- (a) feel that you are an accessible administrator, and
- (b) are aware of the schools programs and accomplishments.

How do you think the general public perceives you and your role in the school? What leads you to these beliefs?

How do you communicate the accomplishments of your teachers and students to your parents?

Are the parents of the students in your school generally supportive? To what factors do you attribute this?

How has this supportive atmosphere affected the basic skills of your students?

In your opinion, how much of the increase in basic skills can be attributed to public support and administrator accessibility?

Other Factors

There may be other factors which you feel have contributed to the increase in CTBS composite score increases in your school during the period between 1955 and 1991 but have not been explored in this interview.

What other factors have contributed to the positive increase in CTBS composite scores during the period between 1985 and 1991?

How have these factors contributed to this increase?







